

DECISION

Fair Work Act 2009 s.185—Enterprise agreement

Independent Education Union of Australia (AG2021/8595)

PARKLANDS CHRISTIAN COLLEGE COLLECTIVE ENTERPRISE AGREEMENT 2021

Educational services

COMMISSIONER SPENCER

BRISBANE, 10 DECEMBER 2021

Application for approval of the Parklands Christian College Collective Enterprise Agreement 2021

- [1] An application has been made for approval of an enterprise agreement known as the *Parklands Christian College Collective Enterprise Agreement 2021* (the Agreement). The application was made pursuant to s.185 of the *Fair Work Act 2009* (the Act). It has been made by the Queensland and Northern Territory Branch of the Independent Education Union of Australia (IEU/the Applicant). The Agreement is a single enterprise agreement.
- [2] Subject to matters that have been addressed by way of undertakings, I am satisfied that each of the requirements of ss.186, 187 and 188 of the Act as are relevant to this application for approval have been met.
- [3] As noted, pursuant to s.190(3), I have accepted undertakings from the employer. In accordance with ss.191(1) and 201(3) of the Act the undertakings are taken to be a term of the Agreement. A copy of the undertakings is attached to the Agreement.
- [4] The IEU being a bargaining representative for the Agreement, has given notice under s.183 of the Act that it wants the Agreement to cover it. In accordance with s.201(2) I note that the Agreement covers the organisation.
- [5] The Agreement is approved and, in accordance with s.54 of the Act, will operate from 17 December 2021. The nominal expiry date of the Agreement is 30 June 2024.

COMMISSIONER

Printed by authority of the Commonwealth Government Printer

<AE514230 PR736659

PARKLANDS CHRISTIAN COLLEGE COLLECTIVE ENTERPRISE AGREEMENT 2021

Subject Matter	Section No.
PART 1 - PRELIMINARY	
Title of Agreement	1.1
Coverage	
Date and Period of Operation	1.3
Agreement Posting	1.4
Single Bargaining Unit	1.5
Definitions	1.6
PART 2 - RELATIONSHIPS TO THE AIMS OF THE SCHOOL	
Mission Statement	2.1
Our Core Values	2.2
School Motto	2.3
Objectives	2.4
College Improvement	2.5
Statement of Faith	2.6
Pastoral Care	2.7
Acknowledgement	2.8
PART 3 - RELATIONSHIP TO INDUSTRIAL INSTRUMENTS	
Relationship to Industrial Instruments	3.1
Relationship to the National Employment Standards	3.2
PART 4 - SALARY & WAGE RELATED MATTERS	
Wage Rates	4.1
Salary Increments	4.2
Acknowledged Teachers	4.3
No Further Claims	4.4
PART 5 - EFFICIENCY AND EFFECTIVENESS INITIATIVES	
Employment Categories - Teachers	5.1
The Working Year	5.2
New Staff Orientation	5.3
Hours of Duty - Teachers	5.4
Reducing School Operating Costs	5.5
Personal Leave	5.6
Special Emergent Leave	5.7
Carer's Leave	5.8
Terminal Illness Carer's Leave	5.9
Parental or Adoption Leave	5.10
Domestic Violence Leave	5.11
Superannuation	5.12
Promoting the School and Teaching Profession	5.13
Union Recognition	5.14
Resourcing School Level Industrial Practice	5.15
Pay Slips	5.16

Workplace Harassment	5.17
Co-Curricular Activities	5.18
Professional Development	5.19
Individual Flexibility	
•	
PART 6 - NON-SALARY BENEFIT ITEMS	
Long Service Leave	6.1
Cashing Out of Long Service Leave	6.2
Proportion of Salary	
, , , , , , , , , , , , , , , , , , , ,	
PART 7 - DISPUTE RESOLUTION AND CONSULTATION	
Dispute Resolution	7.1
Introduction of Changes	
PART 8 - TERMINATION OF EMPLOYMENT & REDUNDANCY	
Termination of Employment	8.1
Redundancy	
PART 9 - SAVINGS CLAUSE	
PARI 9 - SAVINGS CLAUSE	
PART 10 - FUTURE RENEGOTIATION OF AGREEMENT	
Variation, Renewal or Replacement of Agreement	10.1
PART 11 - SIGNATURES	
TAKT TT- GIGNATORES	
SCHEDULE 1 - WAGES, SALARIES AND ALLOWANCES	
SCHEDULE 2 - STATEMENT OF FAITH	
SCHEDULE 3 - CONDITIONS SPECIFIC TO SCHOOL OFFICERS	
SCHEDULE 4 - CONDITIONS SPECIFIC TO SERVICES STAFF	
SCHEDULE 5 – CONDITIONS SPECIFIC TO TAFE TUTORS/TRAINERS	
SCHEDULE 6 - CONDITIONS SPECIFIC TO SOCIAL AND COMMUNITY W	ORKERS

PART 1 - PRELIMINARY

1.1 Title of Agreement

This Agreement shall be known as the Parklands Christian College Collective Enterprise Agreement 2021.

1.2 Coverage

This Agreement shall apply to employees of Parklands Christian College (ACN 097 063 584) and to the Union signatory to this Agreement.

1.3 Date and Period of Operation

- 1.3.1 This Agreement shall commence operation seven (7) days after approval by The Fair Work Commission and shall remain in force until 30 June 2024.
- 1.3.2 Where this Collective Enterprise Agreement specifies an earlier operative date in relation to a particular provision, then that provision shall operate from that date for all staff employed at that earlier date.
- 1.3.3 The Agreement may only be amended or terminated by any of the Parties to the Agreement in the manner prescribed by the *Fair Work Act 2009*.

1.4 Agreement Posting

A copy of this Collective Agreement shall be exhibited in a conspicuous and convenient place at Parklands Christian College so as to be easily read by employees.

1.5 Single Bargaining Unit

For the purposes of negotiating this Collective Agreement a Single Bargaining Unit (SBU) has been established consisting of Employee, Union and Management representatives.

1.6 Definitions

- 1.6.1 The "Act" means the Fair Work Act 2009 as amended or replaced from time to time.
- 1.6.2 "Approved" used in relation to course, diploma, degree, or university, means a course, diploma, degree or university acceptable to be recognised by the Principal and supporting Teacher training Colleges named as Christian Heritage College and Alphacrucis College and Accredited Universities for similar qualifying purposes in respect of teachers employed by Parklands Christian College.
- 1.6.3 "Extra-Curricular Duties" includes the supervision of school sports, clubs within the school, cadet training, community care, missions and other like duties.
 - Such duties shall be voluntary and honorary and shall be arranged by agreement between the Teachers and the employer.
- 1.6.4 "Duty" includes all tasks related to the educational development of students and tasks concerned with the maintenance of good order and behaviour.

Such tasks include class and group teaching and instruction; conducting excursions for instruction in school curriculum during school hours; setting and correcting assignments, tests, laboratory work, and examination papers; supervision of tests and examinations conducted in connection with the school curriculum; evaluating and assessing student work; care of laboratories and their equipment; compiling

mark registers; compiling syllabuses and work books; giving written reports on students' work and progress; and supervising students for behaviour purposes.

- 1.6.5 "Employer" means, Parklands Christian College.
- 1.6.6 "FWC" means, the Fair Work Commission.
- 1.6.7 "NES" means, the National Employment Standards.
- 1.6.8 "Party" means, for the purposes of this Agreement, an employer, an employee or a union which is covered by this Agreement.
- 1.6.9 Head of Faculty means a Teacher appointed as such by the controlling authority of the school, and who is responsible for:
 - (a) prescription of the subject or subject area syllabus with special attention to grading and progression;
 - (b) planning of programs for year and term;
 - (c) arrangement of subject discussions and demonstrations;
 - (d) general guidance of Teachers of the subject with particular attention to young Teachers;
 - (e) enrichment of other modification of subject prescription to suit student groups;
 - conduct of examinations and the planning or organising of a program of continuous assessment in the subject;
 - (g) ordinary maintenance of books and equipment;
 - (h) direction of remedial work in the subject
- 1.6.10 "Registered Teacher" means a person appointed as a Teacher who holds an Approved degree from a recognised tertiary education institution or an Approved equivalent tertiary qualification plus at least one Year of Teacher education or such other qualifications recognised by the Employer as equivalent to one Year of Teacher education.
- 1.6.11 "Year" used in relation to an Approved university course or an Approved course at an accredited institution means the subject requirements specified for a Year of a full-time course or an equivalent number or point value of subjects which are relevant to subjects being taught.
- 1.6.12 "Teacher" means and includes any employee other than the principal, who is ordinarily engaged in teaching full-time or part time on the staff of a school.
 - The term also includes any Teacher engaged in giving class instruction in physical education, commercial subjects, home science, agricultural subjects, art, music, manual training subjects or such other subjects approved by the Queensland Curriculum and Assessment Authority (QCAA)
- 1.6.13 "Union" means the Independent Education Union of Australia Queensland and Northern Territory Branch or any other organisation which is registered pursuant to the Fair Work (Registered Organisations) Act 2009 and which is eligible to cover an employee to which this Agreement applies.
- 1.6.14 "Year of Service" for the purpose of assessing the salary payable to a Teacher shall be determined with due regard to their qualifications and full time equivalent years of service in

a capacity equivalent to the particular category of Teacher in a school or schools to which this Agreement applies.

PART 2 - RELATIONSHIP TO THE AIMS OF THE SCHOOL

2.1 Mission Statement

Parklands Christian College's mission is to serve with Christian purpose, the local community by providing facilities and resources for the education of children that is characterised by:

- (a) striving for academic standards appropriate for each child:
- (b) developing individuality, creativity and critical thinking in students;
- (c) training students in the moral standards of the Bible and the life of Jesus as revealed in the Gospel stories;
- (d) leading students to accept Jesus Christ as their Lord and Saviour;
- (e) fostering self-discipline, respect and love for one another;
- (f) recognising the contribution of Parents by involving them, where possible in Parklands Christian College's learning program;
- (g) valuing Christian Teachers by developing their gifts and encouraging their participation in the life of Parklands Christian College.

2.2 Our Core Values

- (a) Christlikeness
- (b) Learning
- (c) Excellence
- (d) Attitude
- (c) Respect

2.3 School Moto

Foundations for Life.

2.4 Objectives

- 2.4.1 The parties acknowledge that the primary purpose of the College is to provide, at a reasonable cost and primarily for children from Christian homes, an education with both academic rigour and care for children and families through Christian purpose.
- 2.4.2 This Agreement provides a framework for management and staff to work together towards providing and improving care, productivity and efficiency in order to promote the values of our school.
- 2.4.3 This agreement provides a framework for management, employees and union to work together towards improving productivity and efficiency.
- 2.4.4 The Objectives of the Enterprise Agreement are to develop a culture of workplace reform

by:

- (a) providing quality Christian education which meets the expectations of students, parents and staff and Statutory Authorities;
- (b) providing excellence in teaching and encouraging participation in appropriate professional development;
- (c) providing a quality education with the goal of achieving optimum learning outcomes for all students;
- (d) providing a working environment based on the Christian ethos and mutual respect and co-operation between management and staff;
- improving the effectiveness of support services in contributing to the provision of quality teaching and learning and the pastoral care of all students and staff;
- (f) enhancing the provision of broad curriculum services for students through effective and efficient School arrangements and practice in keeping with our Christian ethos;
- (g) developing collaborative and consultative decision-making processes to the achievement of a common goal;
- (h) creating more effective College management and resourcing processes which increase flexibility, responsiveness and professionalism;
- (i) defining and recognising the key role of staff members in Parklands Christian College and making every possible provision for fair and equitable employment principles;
- reviewing resources to ensure cost minimisation without affecting quality or standards.

2.5 College Improvement

The parties agree that over the life of this agreement, staff will devise and support methods of achieving improvement in all the following areas, to the point where the staff and students see the daily achievement of these as part of the culture of the college:

- (a) ensuring the security of classrooms after hours and at breaks. Classroom doors and windows shall be locked at the end of each day;
- (b) maintaining classrooms and grounds to a high standard of neatness and presentation;
- (c) maintenance of a litter-free environment;
- (d) the vigilant observance of a vandalism/graffiti free environment;
- developing community within the College by encouraging parental participation in the education process; Lights, fans and air conditioning shall be turned off at the end of each day; and
- staff shall endeavour to minimize the amount of photocopying.

2.6 Statement of Faith

Employees at the College accept the Statement of Faith listed below and in Schedule 2

- (a) We believe that the Bible is the inspired and infallible Word of God;
- (b) We believe in one God, eternally existent in three persons. Father, Son and Holy Spirit;
- (c) We believe in the Lord Jesus Christ, the second person of the triune Godhead, who was and is the eternal Son of God, that he became incarnate by the Holy Spirit and was born of the virgin Mary.
- (d) We believe in his sinless life, miraculous ministry, substitutionary, atoning death, bodily resurrection, glorious ascension, and abiding intercession.
- (e) We believe that he will come again to this earth at the end of the age to judge the living and the dead and to reign as King of kings and Lord of lords;
- (f) We believe that salvation is by grace through faith in the Lord Jesus Christ; that it is a work of the Holy Spirit whereby the repentant and believing sinner is born again and becomes a new creation in Christ Jesus;
- (g) We believe in the sanctifying power of the Holy Spirit which enables a believer to live a holy and God glorifying life.

2.7 Pastoral Care

All staff should display in every aspect of their daily performance, true Christian virtue and care. If a lack of care is observed in a staff member the matter will be addressed with that individual.

2.8 Acknowledgement

- 2.8.1 The College Management and employees acknowledge and work towards the achievement of the school mission detailed in Clause 2.1.
- 2.8.2 As part of this acknowledgment the Parties agree that attendance at devotions and briefing sessions, whilst honorary and voluntary, will be part of the normal routine of Parklands Christian College.

PART 3 - RELATIONSHIP TO INDUSTRIAL INSTRUMENTS

3.1 Relationship to Industrial Instruments

This agreement shall form the complete agreement covering all terms and conditions of employment and it shall operate to the exclusion of any and all other legally binding agreements, awards or any other industrial instruments whatsoever.

3.2 Relationship to the National Employment Standards

The provisions of the National Employment Standards (NES) are incorporated into this Agreement and except as varied by this Agreement are deemed to have application. Nothing in this Agreement shall be deemed to diminish an employee's entitlement under the NES, except where such variation is permitted under the terms of the NES.

PART 4 - SALARY & WAGE RELATED MATTERS

4.1 Wage Rates

This Agreement provides for the following wage increases as outlined in Schedule 1.

- 4.1.1 This Agreement provides for the following wage increases for teaching staff:
 - (a) With effect from the 1 January 2021, teachers shall receive a 2.5% salary increase from 1 January 2021 as outlined in Schedule 1.
 - (b) With effect from the 1 January 2022, teachers shall receive a 2.5% salary increase from 1 January 2022 as outlined in Schedule 1.
 - (c) With effect from the 1 January 2023, teachers shall receive a 2.5% salary increase from 1 January 2023 as outlined in Schedule 1.
 - (d) With effect from the 1 January 2024, teachers shall receive a 2.5% salary increase from 1 January 2024 as outlined in Schedule 1.
- **4.1.2** The method of negotiating such a percentage pay increase will be by means of the Single Bargaining Unit.

4.2 Salary Increments - Teachers

Except as otherwise provided in this Agreement and subject to satisfactory conduct, diligence, and efficiency, a Teacher shall receive annual increments in salary according to the scale of salaries applicable until the Teacher receives the maximum salary for which the Teacher is eligible under this Agreement.

4.3 Acknowledged Teachers

4.3.1 Advanced Teacher

Consideration will be given to the recognition of Advanced Teachers during the life of this agreement. Applicants can be confidentially nominated by a Head of School, or can make the Principal aware of their wish to apply for the position of Advanced Teacher. This application should be made in writing and addressed to both the Principal and the Head of School. After the initial application, made by the employee, a discussion will occur between appropriate parties to determine whether the applicant fits the criteria of an Advanced Teacher.

- (a) It would be for a 2-year tenure, with annual review by HOS and Principal. It would be renewable on the outcome of these reviews.
- (b) Applications from teachers below Band 3 Step 4 can be considered, with a minimum start at Band 2 Step 3. In these cases, the salary benefit will be equal to the difference between the Band 3 Step 4 and the Advanced Teacher pay scale.
- (c) Advanced Teachers would have no loss of classroom time.

4.3.2 Master Teacher

Consideration will be given to recognition of Master Teacher during the life of this agreement. Applicants can be confidentially nominated by a Head of School, or can make the Principal aware of their wish to apply for the position of Master Teacher. This application should be made in writing and addressed to both the Principal and the Head of School. After the initial application, made by the employee, a discussion will occur between

appropriate parties to determine whether the applicant fits the criteria of a Master Teacher.

- It would be for a 2-year tenure, with annual review by HOS and Principal. It would be (a) renewable on the outcome of these reviews.
- It would only be available to the Band 3 Step 4 and would have no loss of classroom (b) time.

4.4 No Further Claims

There shall be no further wage increases during the life of this Agreement other than those referred to in Schedule 1.

PART 5 - EFFICIENCY AND EFFECTIVENESS INITIATIVES

5.1 **Employment Categories - Teachers**

Teachers shall be advised in writing of their employment category upon appointment.

(a)	Full Time Teacher	Full year employment, every-day, ongoing
(b)	Part-time Teachers	Full year employment, some days in week, ongoing. A part-time teacher will be paid the hourly rate (as outlined in Schedule 1). The hourly rate will be calculated by dividing the fortnightly rate of pay by 62 hours.
(c)	Casual Teachers	Short term, daily needs only. A casual teacher will be paid the casual hourly rate (as outlined in Schedule 1) with a minimum engagement of half a day. The casual hourly rate will be calculated by dividing the fortnightly rate by 62 and adding a 25% casual loading.
(d)	Fixed Term Appointment	Full or Part time, for a set time and task

5.2 The Working Year

- Non-Teaching staff will be entitled to twenty (20) days annual leave pro-rata except where special negotiated provisions apply which must, in part, cover the period of compulsory shut down time of two (2) weeks which include Christmas and New Year holidays unless otherwise arranged.
- Teaching staff will be entitled to a minimum of twelve (12) weeks leave per year in the allocated slots of student breaks. These are six (6) weeks in December to January, two (2) weeks at end of Term 1 which may overlap with Easter, two (2) weeks mid-year, and two (2) weeks in September and October.
- 5.2.3 Teaching staff will commence each teaching term, planned and ready to teach in accordance with College procedures.
- There will be a minimum of eleven (11) pupil-free days during the working year which may be used for but not limited to the following:
 - learning space preparation prior to the commencement of the school year by students; (a)
 - curriculum development in-service, professional development and planning; (b)

- (c) finalisation of reporting and completion of the formalities of the academic year after the students complete the school year;
- (d) The Retreat for staff.

5.2.5 Annual Leave Loading - Employees other than Teachers

- (a) In respect to annual leave entitlements to which clause 5.2 applies, annual leave pay (including any proportionate payments) shall comprise:
 - (i) the employee's ordinary wage rate as prescribed by the Schedule 1 for the period of the annual leave; and
 - (ii) a further amount calculated at the rate of 17.5% of the amounts referred to in clause 5.2.4 (a) (i).
- (b) Clause 5.2.4 (a) shall not apply to:
 - (i) any period or periods of leave exceeding 4 weeks per annum in any other case.
 - (ii) employers (and their employees) who are already paying (or receiving) an annual leave bonus, loading or other annual leave payment which is not less favourable to employees.

5.2.6 Annual Leave Loading - Teachers

- (a) A Teacher who has taught (or has been granted leave by the school) for each day of the school year at the particular school shall receive an annual leave loading equivalent to 17.5% of 4 weeks' salary calculated upon the salary which such employee was receiving immediately before commencing the midsummer vacation.
- (b) A Teacher who commences employment after the beginning of a school year and:
 - (i) has actually taught for at least 20 weeks; or
 - (ii) has taught for at least a full school term and who teaches to the end of the school year,

shall be paid the proportion of the annual leave loading prescribed in clause 5.2.5 (a) that the Teacher's service (excluding school vacations) bears to a standard school year.

- (c) A Teacher who resigns, having given the prescribed notice in writing, or whose services are terminated by the employer for some reason or reasons other than misconduct and who has taught for at least the full first school term shall be paid the proportion of the annual leave loading prescribed in clause 5.2.5 that the Teacher's service (excluding school vacations) bears to a standard school year:
 - Provided that such loading shall be calculated upon salary which the employee was receiving immediately before cessation of employment.
- (d) The full amount of the abovementioned annual leave loadings shall be paid to the Teacher at the commencement of the midsummer vacation or prior date of cessation of employment.
- (e) A standard year shall be deemed for the purposes of clause 5.2.6 to be 40 weeks.

(f) For the purposes of clause 5.2.6, salaries shall be regarded as including allowances prescribed by clause 4.3.

5.3 New Staff Orientation

Teaching staff who commence their service with Parklands Christian College at the beginning of the school year shall undertake an orientation program three (3) days prior to the commencement of the preparation program for continuing staff. The orientation program shall include:

- (a) Letters of appointment including wages/salary and classification.
- (b) Provision of a duty statement for appointment which includes, but is not limited to hours of duty.
- (c) Identification of lines of support.
- (d) Provision of material relevant to the ethos and mission of the school/system.
- (e) Provision of documents relevant to the school/system policies and procedures.
- (f) Provision of information relevant to superannuation entitlements and options.
- (g) Provision of this Agreement.
- (h) Information about income protection.
- (i) Provision of information relevant to union coverage.
- (j) Identification of placement of industrial and occupational health and safety documents and their availability to staff.
- (k) A copy of the current school induction program will be provided to each new employee prior to their induction and training provided to persons responsible for the induction of new employees.

5.4 Hours of Duty – Primary and Secondary Teachers

5.4.1 Contact Time

(a) Maximum contact hours will be as follows:

(i) Secondary:

25 hours

(ii) Primary:

25 hours

- (b) Contact time includes programmed teaching and programmed sport.
- (c) Minimum preparation and correction time will be two (2) hours.
- 5.4.2 Teaching Staff shall be provided with fifty (50) minutes of paid morning tea time per week. Other meal breaks are unpaid.
- 5.4.3 The balance of the thirty-one (31) hours shall be made up by 'other' duties including but not limited to:
 - (a) pastoral care time;
 - (b) playground supervision;

- (c) staff meetings;
- (d) assemblies and chapels;
- (e) parent-teacher conferences;
- (f) sport/cultural supervision; and
- (g) fete.
- 5.4.4 Teaching Staff will additionally perform up to eight (8) hours per semester for such activities as Awards Nights, School Fetes, Evening Parent-Teacher interviewing and the like.

5.5 Reducing School Operating Costs

The Parties agree to examine all areas, over the life of the Agreement to reduce school operating costs. Specific attention will include, but not be limited to, the following areas:

- (a) rationalising of resources both intra and inter campus;
- (b) telephone costs;
- (c) photocopying costs;
- (d) maintenance costs;
- (e) cleaning costs;
- (f) energy costs;
- (g) security;
- (h) internet costs. With consideration to personal uploads/downloads.

5.6 Personal Leave

- 5.6.1 New employees will be granted upon appointment their pro rata entitlement for that calendar year at the rate of two (2) days per term or part thereof.
- 5.6.2 Full time employees are entitled to sick leave accrual at the rate of ten (10) days per year. Such accrual will operate from and include the first pay period after the commencement of the 2001 school year.
- 5.6.3 Sick leave is accumulated at one (1) day for every 5.2 weeks worked.
- 5.6.4 Part-time or term-time employees are entitled to sick leave on a pro rata basis.

5.7 Special Emergent Leave

Employees are entitled to a maximum of three (3) days paid leave per annum non-accumulative for special circumstances or emergencies such as home damage - fire, flood, burglary etc. This leave entitlement is granted on a case by case basis.

5.8 Carers Leave

An employee may access accrued sick leave entitlements to provide care and support for members of his or her immediate family or household when they are ill.

5.9 Terminal Illness Carers Leave

- 5.9.1 Employees are entitled to access accrued leave to care and support a household or an immediate family member who is terminally ill. A further three (3) months of unpaid leave can be accessed by the employee to continue such care and support if necessary.
 - 5.9.2 The College may request a doctor's certificate indicating the terminal nature of the illness.

5.10 Parental or Adoption Leave

- 5.10.1 An employee who has completed at least twelve months continuous service is entitled to unpaid parental leave in accordance with the provisions of the Fair Work Act 2009.
- 5.10.2 An employee may access parental leave entitlements to prepare and care for their infant baby for a period of up to 52 weeks based on the following conditions:
 - (a) For 1- 2 years of continuous employment paid leave for 6 weeks
 - (b) For 3- 4 years of continuous employment paid leave for 8 weeks
 - (c) For 5+ years of continuous employment paid leave for 12 weeks
- 5.10.3 A total of 12 months leave, the remainder after paid leave being unpaid leave. An employee is eligible to access a further 52 weeks unpaid leave as prescribed in the NES.
- 5.10.4 A return to employment at the end of the period but not necessarily to the same job if the operating requirements of the school have changed.
- 5.10.5 If an employee's spouse has given birth to a child, the employee is entitled to a period of ten days paid paternity leave, which must be taken within 12 weeks of the date of confinement.
- 5.10.6 An employee will be entitled to any government funded leave, in addition to the employer funded days.
- 5.10.7 The period of leave nominated by the employee will be taken at the time of the confinement, or immediately thereafter, adoption or to make arrangements for the care of other children in the family.

5.11 Domestic Violence Leave

5.11.1 Leave

- (a) An employee, who is experiencing domestic violence, will have access to ten (10) days per year non-cumulative of paid special leave in order to address related matters including, but not limited to:
 - (i) attending medical and/or counselling appointments;
 - (ii) sourcing alternative accommodation;
 - (iii) accessing legal advice;
 - (iv) attending legal proceedings;
 - (v) organising alternative care for members of their immediate family or household;

- (vi) organising alternative education arrangements for their children;
- (vii) rebuilding support networks; and
- (viii) other issues related to the domestic violence.
- (b) This leave will be in addition to existing leave entitlements and may be taken as consecutive or single days or as a fraction of a day and all reasonable requests will be approved.
- (c) Employees can also access existing leave entitlements for the abovementioned purposes, without the usual notice requirements.
- (d) It is not mandatory for the employee to have exhausted other forms of paid leave prior to accessing this special leave.

5.11.2 Supporting Another Person Experiencing Domestic Violence

- (a) An employee who supports a person experiencing domestic violence may use their existing carer's leave to accompany the person on activities related to that personal crisis, or to mind the children of the person to enable them to undertake activities related to such significant matter.
- (b) This sub-clause applies only where an employee supports a person who is a member of their immediate family or household.

5.12 Superannuation

Application - In addition to the rates of pay prescribed by this Agreement, eligible employees, as defined in clause 5.12.1 (b), shall be entitled to Superannuation Benefits, in accordance with the provisions of this clause 5.12.

5.12.1 Definitions

- (a) "Preferred Fund" means the employee's preferred Superannuation provider.
- (b) "Eligible employee" means an employee who has earned \$450 or more in any month until 1 July 2022 then all employees.
- (c) "Ordinary time earnings" means the actual ordinary rate of pay the employee receives for ordinary hours of work including shift loading, skill allowances and supervisory allowances where applicable. The term includes any over-award payment as well as casual rates received for ordinary hours of work. Ordinary time earnings shall not include overtime, disability allowances, commission, bonuses, lump sum payments made as a consequence of the termination of employment, annual leave loading, penalty rates for public holiday work, fares and travelling time allowances or any other extraneous payments of a like nature.

5.12.2 Contributions

Amount - the employer shall contribute (as a minimum) on behalf of each eligible employee (and current employees irrespective of age) an amount calculated at the appropriate Superannuation Guarantee rate of the employee's ordinary time earnings, into an Approved Fund. Each such payment of contributions shall be rounded off on an equitable basis either to the nearest cent or 10 cents provided that there is consistency of application to all employees of the employer. The superannuation guarantee charge (SGC) shall increase in line with the Superannuation Guarantee Charge Act 1992, as amended from time to time.

(a) Regular Payment - The employer shall pay such contributions to the credit of each such employee at least once each calendar month or in accordance with the requirements of the superannuation Fund Trust Deed.

- (b) Minimum level of earnings Until 1 July 2022 an employer will not be required to pay superannuation contributions on behalf of any eligible employee whether full-time, part-time, casual, adult or junior in respect of any month where the employee's earnings are less than \$450.
- (c) Absences from work- contributions shall continue to be paid on behalf of an eligible employee during any absence on paid leave. Except in the case of absence on Workers' Compensation an employer will not be required to pay superannuation contributions on behalf of any eligible employee during any period of leave without pay. In the case of Workers' Compensation, the employer will contribute 3% of the employee's ordinary time earnings whenever the employee is receiving by way of Workers' Compensation, an amount of money no less than the Award rate of pay.
- (d) Where an employee wishes to make voluntary contributions to superannuation, the employee may authorise the company to deduct from the employee's wages an amount specified by the employee.
- (e) Additional employee contributions and/or additional employer contributions, in accordance with the provisions of this Agreement, will be forwarded to the Fund by the employer at the same time as the employer's contributions made under clause 5.12.2 (b).
- (f) Cessation of Contributions An employer shall not be required to make any further contributions on behalf on an eligible employee for any period after the end of the ordinary working day upon which the contract of employment ceases to exist.
- (g) No Other Deductions No additional amounts shall be paid by the employer for the establishment, administration, management or any other charges in connection with the Fund other than the remission of contributions as prescribed herein.

5.12.3 Salary Sacrifice to Superannuation

An employee may elect to Salary Sacrifice an amount to Superannuation. Where the employee so elects the following provisions will apply:

- (a) The employer will continue to calculate the contributions required by clause 5.12.2 (a) and/or the Superannuation Guarantee (Administration) Act 1992 on the basis of the employee's ordinary time earnings before the salary sacrifice is deducted.
- (b) Salary sacrifice deductions will be made during a period of paid leave and the employee will receive the rate of pay specified under this agreement less the salary sacrifice deduction.
- (c) Calculation of salary for all purposes including leave accruals and other payments due on termination of employment shall be calculated on a rate of pay which includes the salary sacrifice contributions.

5.12.4 Enrolment

- (a) The employer shall as soon as practicable as to both current and future eligible employees:
 - (i) Notify each employee of the employee's entitlement to Superannuation.
 - (ii) Notify each employee to join, if applicable, an Approved Fund online and provide the School with their Membership number.

- (b) Each employee upon becoming eligible to become a member of the superannuation Fund will:
 - (i) complete the sign-up process with their elected Superannuation fund to enable that employee to become a member of the Fund; and
 - (ii) return the Membership number to the employer within 28 days of receipt.
- (c) Notwithstanding the failure of an employee to complete the required paperwork, the employer will make payments on behalf of the employee to the Superannuation fund.
- (d) Where an employee has not exercised a choice in relation to their preferred fund identified within 28 days of commencing employment, the employer will make superannuation payments to the school's default fund which is Christian Super.

5.12.5 Unpaid Contributions

Subject to the *Industrial Relations Act* where it has been established that the employer has failed to comply with the requirements of clause 5.12.1 (b) in respect of any eligible employee, such employer will be liable to make the appropriate contributions retrospectively to the date of eligibility of the employee, plus an amount equivalent to the rate of return those contributions would have attracted in the Superannuation Fund had they been paid on the due dates.

The making of such contributions satisfies the requirements of clause 5.12 excepting that resort to the provisions of clause 5.12.6 shall not limit any common law action which may be available in relation to death, disablement or any similar cover existing within the terms of the Superannuation Fund.

5.12.6 Amendments to this Clause

In the event that the Fair Work Commission determines to generally amend the definition of Ordinary Time Earnings or the provisions around payment of superannuation whilst on WorkCover such amended provisions will be inserted into this Agreement. This commitment is subject to the provisions of the Act regarding amendment of Agreements.

5.13 Promoting the School and the Teaching Profession

- 5.13.1 The parties agree to work towards improving the status of the teaching profession in the Community and to increase the attractiveness of the teaching profession to the students and parents of Parklands Christian College. This will be both in word and deed. All staff are expected to be aware that they are representing the school and its staff and as such should uphold the school in all comments made to parents and students and to dress in the uniform prescribed by the school appropriate to the day's activities and function.
- 5.13.2 Staff undertake to give attention to their own commitment to the profession and maintain an active awareness of the issues and directions of learning and teaching.

5.14 Union Recognition

- 5.14.1 The College recognises the Unions' party to this Agreement as the legitimate industrial representatives of the employees according to their Unions' respective callings.
- 5.14.2 The College and the Union are committed to working constructively together to further the interests of education and specifically non-governmental education and the employees in this industry, while at the same time working constructively together to further the interests of Parklands Christian College, as expressed in this agreement and other public

documents.

5.15 Resourcing School Level Industrial Practice

5.15.1 Resourcing in the Workplace

- (a) The employer recognises that the union's party to this Agreement and their accredited representatives are the legitimate representatives of employees covered by this Agreement and shall in no way discourage employees to join or maintain membership of the relevant unions bound by this Agreement.
- (b) The employer recognises the value of employees gaining a better understanding of industrial relations within the employer's operations, and in particular, a better understanding of the industrial instruments and industrial issues impinging upon the working life of employees and the enterprise bargaining process.
- (c) The employer agrees to provide:
 - adequate time during normal working hours for workplace union representatives to carry out union responsibilities and this time shall be recognised in the employees' designated hours of duty;
 - (ii) access for workplace union representatives to telephone, fax, photocopier, computer email and internet to carry out union responsibilities and shall not deny email/fax access to the unions party to this Agreement for the forwarding of documents to employees;
 - (iii) reasonable time for workplace union representatives to consult with employees on workplace matters and enterprise bargaining issues;
 - (iv) reasonable opportunity for workplace union representation to provide union information to new employees;
- (d) One month's notice must be given of the employee's intention to take this leave.
- (e) The application for leave shall be endorsed by the respective union.
- (f) The granting of such leave shall be subject to the reasonable convenience of the employer having regard to the efficient operation of the College.
- (g) No employee shall be granted leave in excess of the duration of the course i.e. travelling time is not included.
- (h) The employer is not responsible for any additional costs except for the payment of relief employees where this is considered appropriate.

5.15.2 Industrial Training

The employer will provide leave with pay of up to three (3) days per annum for IEU-QNT members to undertake courses provided by the Independent Education Union - Queensland and Northern Territory Branch (IEU-QNT) or a body authorised by IEU-QNT to do so on its behalf.

5.16 Pay slips

The employer shall provide all employees with Pay Advice Slips showing the following:

- (a) salary band and step/classification level;
- (b) gross payment;
- (c) tax paid;
- (d) net payment;
- (e) superannuation contributions.

5.17 Workplace Harassment

5.17.1 Preamble

- (a) The parties acknowledge that workplace harassment can occur in a workplace and have a negative effect upon employees in a workplace. Such unacceptable practice impacts on the school community.
- (b) Consequently the College agrees to develop, in conjunction with employees and their representatives a workplace harassment policy and procedures.

5.17.2 Policy

The policy will:

- (a) Define workplace harassment and provide examples of the types of behaviour which constitute such harassment.
- (b) State that workplace harassment is unacceptable and will not be tolerated.
- (c) Include a statement as to the negative impact on individuals, colleagues and the organisation.
- (d) Encourage workers who experience or witness workplace harassment to engage in procedures to end such behaviour.
- (e) Contain a commitment to education and training in regard to the policy and procedures on a regular basis or at least once per year.
- (f) Provide for the appointment, training and time release of contact person(s) to handle complaints.
- (g) Be available to all staff and displayed in prominent places in the school.

5.17.3 Procedures

The procedures will:

- (a) Be fair and equitable.
- (b) Ensure principles of natural justice are upheld.
- (c) Ensure privacy and confidentiality.

- (d) Be undertaken with discretion as to protect the reputation of the person being investigated.
- (e) Be aimed at resolving the problem rapidly.
- (f) Ensure that accurate records and documentation are kept.
- (g) Include informal procedural steps for dealing with the alleged harassment.
- (h) Include formal steps for dealing with the alleged harassment which incorporates an investigative process outlining how and who will conduct the investigation, the rights of both the alleged perpetrator and the alleged victim to representation and the need for each party to receive a report on the outcome.

5.18 Extra-Curricular Activities

- 5.18.1 It is recognised that a wide variety of extra-curricular activities contribute to the mission and ethos of the school and may enhance the holistic development of the student. Therefore, an organised extra-curricular program is provided.
- 5.18.2 Teachers contribute to the ethos and mission of the school by accepting the performance of co-curricular activities which are an integral part of the total school curriculum. The College seeks to formalise more equitable practices in relation to extra-curricular activities. It is the College's view that such practices may generally involve participation in at least two extra-curricular activities during each year.
- 5.18.3 Clause 7.2 of this Agreement shall be considered by the parties in making decisions around co-curricular activities such as:
 - (a) supervision of school sports;
 - (b) clubs within the school; and
 - (c) music and drama productions and other like duties.
- 5.18.4 Extra-curricular activities thus defined shall continue to be performed on an honorary and voluntary basis.

5.19 Professional Development

- 5.19.1 The Principal actively supports a school wide professional development program in consultation with the Teachers, School Officers and Support Staff.
- 5.19.2 Professional development opportunities are advertised on QCAA and ISQ websites which staff should monitor.
- 5.19.3 Professional development held during pupil free days and staff meetings, conducted by the employer for teaching staff will, wherever relevant; articulate the AITSL Australian Professional Standards for Teachers.
- 5.19.4 Teachers attending planned professional development activities should obtain a record of attendance as evidence of training as required for reporting to the Queensland College of Teachers.

5.20 Individual Flexibility

- 5.20.1 An employer and employee covered by this enterprise agreement may agree to make an individual flexibility arrangement to vary the effect of terms of the agreement if:
 - (a) the agreement deals with one (1) or more of the following matters:
 - (i) arrangements about when work is performed;
 - (ii) overtime rates;
 - (iii) penalty rates;
 - (iv) allowances;
 - (v) leave loading; and
 - (b) the arrangement meets the genuine needs of the employer and employee in relation to 1 or more of the matters mentioned in paragraph (a); and
 - (c) the arrangement is genuinely agreed to by the employer and employee.
- 5.20.2 The employer must ensure that the terms of the individual flexibility arrangement:
 - (a) are about permitted matters under section 172 of the Fair Work Act 2009; and
 - (b) are not unlawful terms under section 194 of the Fair Work Act 2009; and
 - (c) result in the employee being better off overall than the employee would be if no arrangement was made.
- 5.20.3 The employer must ensure that the individual flexibility arrangement:
 - (a) is in writing; and
 - (b) includes the name of the employer and employee; and
 - (c) is signed by the employer and employee and if the employee is under 18 years of age, signed by a parent or guardian of the employee; and
 - (d) includes details of:
 - the terms of the enterprise agreement that will be varied by the arrangement;
 and
 - (ii) how the arrangement will vary the effect of the terms; and
 - (iii) how the employee will be better off overall in relation to the terms and conditions of his or her employment as a result of the arrangement; and
 - (e) states the day on which the arrangement commences.
- 5.20.4 The employer must give the employee a copy of the individual flexibility arrangement within 14 days after it is agreed to.
- 5.20.5 The employer or employee may terminate the individual flexibility arrangement:
 - (a) by giving no more than 28 days written notice to the other party to the arrangement;

(b) if the employer and employee agree in writing - at any time.

PART 6 - NON-SALARY BENEFIT ITEMS

6.1 Long Service Leave

- 6.1.1 All employees will accumulate Long Service leave entitlement at the rate of 1.3 weeks per full time equivalent year of completed service from the date of engagement.
- 6.1.2 Employees who complete ten years of continuous service may access their entitlement to Long Service Leave. After the first 10 years, Long Service Leave may be taken at five year intervals, with leave accruing at the same rate, namely 1.3 weeks per full time equivalent year of completed service.
- 6.1.3 Long Service Leave will be offered on a pro rata basis after seven (7) years upon resignation, death, retrenchment or total and permanent incapacity.
- 6.1.4 Timing of leave will be at a mutually convenient time with reasonable notice given. Reasonable notice should be not less than two full school terms.
- 6.1.5 In accordance with the *Industrial Relations Act 1999 (Queensland)* and subject to 6.1.2, long service leave may be taken at a time agreed between the employee and the employer.
- 6.1.6 If an employee fails to agree with the employer on matters, the employer may give the employee at least two full school terms notice of the date on which the employee must take at least four (4) weeks long service leave.
- 6.1.7 The employee must comply with the notice.

6.2 Cashing Out of Long Service Leave

- 6.2.1 The parties agree that long service leave is designed to ensure that long term employees receive a break from service and as such the preference of Parklands Christian College is for employees to access accrued long service as appropriate.
- 6.2.2 The parties accept that under certain circumstances it may be preferable for an employee to request payment of all or part of the long service leave accrual when it falls due without accessing 'absence on leave' from the school. Any application by an employee to have part or all of their accrued long service leave paid out would be subject to the following conditions:
 - (a) the school agrees to make the payment and the arrangement is committed to writing and signed by both the applicant and the school;
 - (b) superannuation at the rate the employee would have received if they had taken the leave would be payable on the cashed-out amount;
 - (c) subject to any requirements at law, the employee will have the right to determine to receive the payment in cash or to salary sacrifice all or part of the amount of the cashing out to superannuation; and
 - (d) the employee completing the relevant form acknowledging the loss of entitlements which the Employee would have otherwise received but for the cashing out of long service leave.

6.2.3 For the purposes of clarity, an agreement to cash out long service leave can only be initiated by the employee and is subject to school approval.

6.3 Proportion of Salary

- 6.3.1 A Teacher upon appointment shall be paid as from the date upon which the Teacher commenced Duty, provided that a Teacher who has taught (or has been granted leave by the school) for each day of the school year at the particular school shall be paid as for a full calendar year commencing on 1st January. A Teacher who ceases Duty before completing 10 teaching weeks of employment shall be paid in lieu of vacation pay an amount equal to 1112th of their ordinary pay for the period of employment.
- 6.3.2 A Teacher who ceases Duty after at least 10 teaching weeks of employment shall be paid the proportion of the Teacher's annual salary of that year that the Teacher's service excluding school vacations bears to a standard school year:
 - Provided that such proportion of salary shall be calculated on the salary which the employee was receiving immediately before cessation of employment.
- 6.3.3 A standard year shall be deemed for the purposes of clause 6.3 to be 40 weeks.

PART 7 - DISPUTE RESOLUTION AND CONSULTATION

7.1 Dispute Resolution

- 7.1.1 If a dispute relates to:
 - (a) a matter arising under the agreement; or
 - (b) the National Employment Standards; or
 - (c) any industrial matter:

this term sets out procedures to settle the dispute.

- 7.1.2 An employee who is a party to the dispute may appoint a representative for the purposes of the procedures in this term. An employee may be represented at any stage of this process.
- 7.1.3 In the first instance, the parties to the dispute must try to resolve the dispute at the workplace level, by discussions between the employee or employees and relevant supervisors and/or management. A response will be provided as soon as reasonably practicable.
- 7.1.4 Where the dispute concerns alleged actions of the immediate supervisor or manager the employee/s may bypass this level (sub clause 7.1.3 above) in the procedure.
- 7.1.5 If the dispute is not resolved under sub clause 7.1.3, the matter may be referred to the next higher level of management for discussion. Such discussion should, if possible, take place within 24 hours after the request has been made.
- 7.1.6 If the dispute involves allegations of discrimination or harassment by a supervisor or manager, the employee may commence this process by reporting the allegations to the next level of management beyond that of the supervisor or manager concerned. If there is no level of management beyond that involved in the allegation, the employee may proceed directly to the process outlined at sub clause 7.1.8.
- 7.1.7 If the dispute is still unresolved after discussions mentioned earlier in this clause, the matter may be raised with the senior management of the employer or the employer's nominated

industrial representative. This should occur as soon as it is evident that the earlier will not result in resolution of the dispute.

- 7.1.8 If discussions at the workplace level do not resolve the dispute, a party to the dispute may refer the matter to the Fair Work Commission.
- 7.1.9 The Fair Work Commission may deal with the dispute in two (2) stages:
 - (a) The Fair Work Commission will first attempt to resolve the dispute as it considers appropriate, including by mediation, conciliation, expressing an opinion or making a recommendation; and
 - (b) if the Fair Work Commission is unable to resolve the dispute at the first stage, The Fair Work Commission may then:
 - (i) arbitrate the dispute; and
 - (ii) make a determination that is binding on the parties.

Note If the Fair Work Commission arbitrates the dispute, it may also use the powers that are available to it under the Act.

A decision that the Fair Work Commission makes when arbitrating a dispute is a decision for the purpose of Div 3 of Part 5.1 of the Act. Therefore, an appeal may be made against the decision.

- 7.1.10 While the parties are trying to resolve the dispute using the procedures in this clause:
 - (a) an employee must continue to perform his or her work as he or she would normally unless he or she has a reasonable concern about an imminent risk to his or her health or safety; and
 - (b) an employee must comply with a direction given by the employer to perform other available work at the same workplace, or at another workplace, unless:
 - (i) the work is not safe; or
 - (ii) applicable occupational health and safety legislation would not permit the work to be performed; or
 - (iii) the work is not appropriate for the employee to perform; or
 - (iv) there are other reasonable grounds for the employee to refuse to comply" with the direction.
 - (c) The status quo existing before the emergence of the grievance or dispute is to continue.
- 7.1.11 The parties to the dispute agree to be bound by a decision made by the Fair Work Commission in accordance with this term.
- 7.1.12 So as to remove doubt, the parties record that the reference to disputes or grievances in respect to any industrial matter includes disputes or grievances in relation to whether the employer had reasonable business grounds for refusing a request under the National Employment Standards for flexible working arrangements or an application to extend unpaid parental leave.

7.2 Introduction of Changes

- 7.2.1 This term applies if the employer:
 - has made a definite decision to introduce a major change to production, program, organisation, structure or technology in relation to its enterprise that is likely to have a significant effect on the employees; or
 - (b) proposes to introduce a change to the regular roster or ordinary hours of work of employees.
- 7.2.2 For a major change referred to in 7.2.1a.:
 - the employer must notify the relevant employees of the decision to introduce the major change; and
 - (b) sub clauses 3 to 9 apply.
- 7.2.3 The relevant employees may appoint a representative for the purposes of the procedures in this term.
- 7.2.4 If:
 - (a) a relevant employee appoints, or relevant employees appoint, a representative for the purposes of consultation; and
 - (b) the employee or employees advise the employer of the identity of the representative;

the employer must recognise the representative.

- 7.2.5 As soon as practicable after making its decision, the employer must:
 - (a) discuss with the relevant employees:
 - (i) the introduction of the change; and
 - (ii) the effect the change is likely to have on the employees; and
 - (iii) measures the employer is taking to avert or mitigate the adverse effect of the change on the employees; and
 - (b) for the purposes of the discussion-provide, in writing, to the relevant employees:
 - (i) all relevant information about the change including the nature of the change proposed; and
 - (ii) information about the expected effects of the change on the employees; and
 - (iii) any other matters likely to affect the employees.
- 7.2.6 However, the employer is not required to disclose confidential or commercially sensitive information to the relevant employees.
- 7.2.7 The employer must give prompt and genuine consideration to matters raised about the major change by the relevant employees.
- 7.2.8 If a term in this agreement provides for a major change to production, program, organisation, structure or technology in relation to the enterprise of the employer, the requirements set out

in paragraph 2a. and sub clauses 3 and 5 are taken not to apply.

- 7.2.9 In this term, a major change is likely to have a significant effect on employees if it results in:
 - (a) the termination of the employment of employees; or
 - (b) major change to the composition, operation or size of the employer's workforce or to the skills required of employees; or
 - (c) the elimination or diminution of job opportunities (including opportunities for promotion or tenure); or
 - (d) the alteration of hours of work; or
 - (e) the need to retrain employees; or
 - (f) the need to relocate employees to another workplace; or
 - (g) the restructuring of jobs.
- 7.2.10 For a change to regular roster or ordinary hours of work of employees:
 - (a) the employer must notify the relevant employees of the proposed change; and
 - (b) sub clauses 11 to 15 apply.
- 7.2.11 The relevant employees may appoint a representative for the purposes of the procedures in this term.

7.2.12 If:

- (a) a relevant employee appoints, or relevant employees appoint, a representative for the purposes of consultation; and
- (b) the employee or employees advise the employer of the identity of the representative; the employer must recognise the representative.
- 7.2.13 As soon as practicable after proposing to introduce the change, the employer must:
 - (a) discuss with the relevant employees the introduction of the change; and
 - (b) for the purposes of the discussion-provide to the relevant employees:
 - (c) all relevant information about the change, including the nature of the change; and
 - (d) information about what the employer reasonably believes will be the effects of the change on the employees; and
 - (e) information about any other matters that the employer reasonably believes are likely to affect the employees; and
 - (f) invite the relevant employees to give their views about the impact of the change (including any impact in relation to their family or caring responsibilities).
- 7.2.14 However, the employer is not required to disclose confidential or commercially sensitive information to the relevant employees.

7.2.15 The employer must give prompt and genuine consideration to matters raised about the change by the relevant employees.

7.2.16 In this term:

- (a) "relevant employees" means the employees who may be affected by a change referred to in sub clause 1.
- (b) "a change to regular roster or ordinary hours of work" does not include a change by way of the College's educational timetable in respect of academic classes and student activities, which:
- (c) may operate on a term, semester or a school year basis; and
- (d) ordinarily changes between one period of operation and the next; and
- (e) may change during the period of operation.

7.3 Casual conversion

This clause is subject to sections 66A to 66M of the Fair Work Act 2009.

7.3.1 Employer offers

- (a) The employer will assess the eligibility of their casual employees for possible conversion to either ongoing part time of full time.
- (b) The employer must make a written offer to convert a casual employee to permanent employment within 21 days after the employee's 12-month anniversary, if the employee:
 - (i) has been employed by the employer for a period of 12 months beginning the day the employment started;
 - (ii) has worked a regular pattern of hours on an ongoing basis for at least the last six (6) months; and
 - (iii) could continue to work these hours as a full time or part-time employee without significant changes.

The offer must be in writing and be an offer for the employee to convert to:

- (i) full time employment, if the employee's hours worked for at least the last 6 months have been the same as full time hours; or
- (ii) part-time employment (consistent with the employee's regular pattern of hours worked for at least the last 6 months), if the employee's hours worked for at least the last 6 months have been less than full time hours.
- (c) To accept an offer to convert, an employee must respond in writing within 21 days after receiving the offer. If no response is received the employer will assume the employee has declined the offer.

7.3.2 Employer not making an offer

Should the employer decide not to offer the casual employee conversion to part-time or full-time employment, the employer will write to the employee within 21 days after the employee's 12-month anniversary advising them:

- (a) that they aren't making an offer of casual conversion; and
- (b) the reasons for not making the offer.

The employer can only decide not to offer casual conversion if:

- (a) the employee hasn't worked a regular pattern of hours:
 - (i) on an ongoing basis for at least the last six (6) months;
 - (ii) which they could continue working as a full time or part time employee without significant changes; or

(b) The employer has reasonable business grounds for not making an offer. If the employee does not accept the employer's decision not to offer conversion, this will constitute a dispute that will be dealt with under the dispute resolution procedure in clause 7.1. Under that procedure, the employee or the employer may refer the matter to the Fair Work Commission if the dispute cannot be resolved at the workplace level.

7.3.3 Employee requesting casual conversion

- (a) Notwithstanding Clauses 7.3.1 and 7.3.2, a person engaged by the employer as a regular casual employee may request, in writing, that their employment be converted to full-time or part-time employment, subject to meeting the requirements of 7.3.1 (a) to (c) above, and if:
 - (i) you haven't refused a previous offer to become a permanent employee in the last six(6) months;
 - (ii) your employer hasn't told you in the last 6 months that they won't offer you casual conversion on reasonable grounds; and
 - (iii) your employer hasn't already refused a request from you to become a permanent employee based on reasonable grounds in the last 6 months.
- (b) Where the employer refuses a regular casual employee's request to convert, the employer must provide the casual employee with the employer's reasons for refusal in writing within 21 days of the request being made. If the employee does not accept the employer's refusal, this will constitute a dispute that will be dealt with under the dispute resolution procedure in clause 9. Under that procedure, the employee or the employer may refer the matter to the Fair Work Commission if the dispute cannot be resolved at the workplace level.

7.3.4 Accepted offer of conversion

- (a) Where it is agreed that a casual employee will have their employment converted to full-time or part-time employment as provided for in this clause (clX.4), the employer and employee must discuss and record in writing
 - (i) the form of employment to which the employee will convert that is, full-time or parttime employment; and
 - (ii) if it is agreed that the employee will become a part-time employee, the hours and days of the week to be worked.
- (b) The conversion will take effect from the start of the next pay cycle following such agreement being reached unless otherwise agreed.

7.3.5 Employer obligations

- (a) The employer must provide a casual employee, whether a regular casual employee or not, with a copy of the provisions of this subclause within the first 12 months of the employee's first engagement to perform work.
- (b) A casual employee's right to request to convert is not affected if the employer fails to comply with the provision of 7.3.1.

PART 8 - TERMINATION OF EMPLOYMENT AND REDUNDANCY

8.1 Termination of Employment

8.1.1 Statement of Employment

The employer shall, in the event of termination of employment, provide upon request to an employee who has been terminated a written statement specifying the period of employment and the classification or type of work performed by the employee.

8.1.2 Termination by Employer - Full-time Employees

(a) The employer shall give to a full-time employee at least one month's notice during term time in writing of the termination of their services.

- (b) Employees over 45 years of age at the time of the giving of notice, and with more than 5 years' service with the employer, shall be entitled to an additional week's notice to that prescribed in clause 8.1.2 (a).
- (c) Payment in lieu of notice shall be made if the appropriate notice is not given:
 - Provided that employment may be terminated by part of the period of notice specified and part payment in lieu thereof.
- (d) In calculating any payment in lieu of notice the ordinary time rate of pay for the employee concerned shall be used.
- (e) The period of notice in clauses 8.1.2 (a) and 8.1.2 (b) shall not apply in the case of dismissal for misconduct or other grounds that justify instant dismissal.

8.1.3 Termination by Employer - Part-time Employees

(a) The employer shall give to part-time employees the following notice in writing of the termination of their services.

Less than 3 years' service 2 weeks' notice

More than 3 years but less than 5 years 3 weeks' notice

More than 5 years 4 weeks' notice

- (b) In addition to the notice in clause 8.1.3 (a) employees over 45 years of age at the time of the giving of notice and with not less than 2 years continuous service, shall be entitled to an additional week's notice.
- (c) Payment in lieu of notice shall be made if the appropriate notice is not given:
 - Provided that employment may be terminated by part of the period of notice specified and part payment in lieu thereof.
- (d) In calculating any payment in lieu of notice the ordinary time rate of pay for the employee concerned shall be used.
- (e) The period of notice in clause 8.1.3(a) shall not apply in the case of dismissal for misconduct or other grounds that justify instant dismissal, or in the case of casual employees.

8.1.4 Notice of Termination by Employee

- (a) The notice of termination required to be given by an employee shall be the same as that required of an employer: Provided that there shall be no additional notice based on the age of the employee concerned.
- (b) If an employee fails to give notice the employer shall have the right to withhold monies due to the employee with a maximum amount equal one weeks' wages for non-teaching staff and two weeks' wages for teaching staff.

8.2 Redundancy

8.2.1 Consultation Before Termination

This section applies if:

- (a) the employer has made a definite decision to introduce a major change to production, program, organisation, structure, or technology in relation to its enterprise; and
- (b) the change is likely to have a significant effect on employees of the enterprise.
- 8.2.2 The employer must notify the relevant employees of the decision to introduce the major change.
- 8.2.3 The relevant employees may appoint a representative for the purposes of the procedures in this term.
- 8.2.4 The employer must recognise the representative if:
 - (c) a relevant employee appoints, or relevant employees appoint, a representative for the purposes of consultation; and
 - (d) the employee or employees advise the employer of the identity of the representative;
- 8.2.5 As soon as practicable after making its decision, the employer must:
 - (e) discuss with the relevant employees:
 - (i) the introduction of the change; and
 - (ii) the effect the change is likely to have on the employees; and
 - (iii) measures the employer is taking to avert or mitigate the adverse effect of the change on the employees; and
 - (f) for the purposes of the discussion provide, in writing, to the relevant employees:
 - (i) all relevant information about the change including the nature of the change proposed; and
 - (ii) information about the expected effects of the change on the employees; and
 - (iii) any other matters likely to affect the employees.
- 8.2.6 However, the employer is not required to disclose confidential or commercially sensitive information to the relevant employees.
- 8.2.7 The employer must give prompt and genuine consideration to matters raised about the major change by the relevant employees.
- 8.2.8 In this section, a major change is likely to have a significant effect on employees if it results in:
 - (g) the termination of the employment of employees; or
 - (h) major change to the composition, operation or size of the employer's workforce or to the skills required of employees; or

- (i) the elimination or diminution of job opportunities (including opportunities for promotion or tenure); or
- (j) the alteration of hours of work; or
- (k) the need to retrain employees; or
- (I) the need to relocate employees to another workplace; or
- (m) the restructuring of jobs.
- 8.2.9 In this section, relevant employees means the employees who may be affected by the major change.

8.2.10 Transfer to Lower Paid Duties

- (a) Where an employee is transferred to lower paid duties the employee shall be entitled to the same period of notice of transfer as the employee would have been entitled to if the employee's employment had been terminated.
- (b) Employees aged over 45 years of age and who have been continually employed for 2 years or more are eligible for an additional week's notice.

8.2.11 Time Off During Notice Period

- (c) Where a decision has been made to terminate an employee due to redundancy, the employee shall be allowed up to one day's time off without loss of pay during each week of notice for the purpose of seeking other employment.
- (d) If the employee has been allowed paid leave for more than one day during the notice period for the purpose of seeking other employment, the employee shall, at the request of the employer, be required to produce proof of attendance at an interview or the employee shall not receive payment for the time absent. For this purpose, a statutory declaration will be sufficient.

8.2.12 Severance Pay

In addition to the period of notice prescribed for termination in clause 8.1of this Agreement, an employee whose employment is terminated due to redundancy will be entitled to the following amounts of severance pay:

Period of Continuous Service Less than 1 year

Severance Pay (weeks' pay) nil	
1 year but not more than 2 years	4
More than 2 years but not more than 3 years	6
More than 3 years but not more than 4 years	7
More than 4 years but not more than 5 years	8
More than 5 years but not more than 6 years	10
More than 6 years but not more than 7 years	11
More than 7 years but not more than 8 years	13
More than 8 years but not more than 9 years	14
More than 9 years	16

PART 9 - SAVINGS CLAUSE

The existing conditions and accrued entitlements of employees employed at the date of signing shall not be reduced as a result of this Agreement coming into effect.

PART 10 - FUTURE RENEGOTIATION OF AGREEMENT

- 10.1 Variation, Renewal or Replacement of Agreement
- 10.1.1 Subject to satisfactory implementation of this Agreement the Parties agree to reopen negotiations by 1 April 2019 with a view to negotiating a replacement Agreement. Further the Parties agree to monitor the implementation of the Agreement through the SBU and identify issues suitable for negotiation in a replacement agreement.
- 10.1.2 The Parties agree that this Collective Agreement may be varied by a further Collective Agreement in circumstances where all of the Parties genuinely agree that a variation is necessary.
- 10.1.3 Any agreed variation to this Collective Agreement will be subject to the same consultation and approval process as that used for the Collective Agreement.

PART 11 - SIGNATURES

Signed for and on behalf of Parklands Christian College	51
ACN 097 063 584	GARY RAMONO CULLY
11 Hillcrest Road PARK RIDGE QLD 4125	(Full Name) PRENCEPAL
In the presence of -	(Signature) (Signature) (Signature) (Signature) (Full Name) BUSINESS MANACOR (Position) 2674 NOVEMBER 2021 (Date)
Signed for and on behalf of the Independent Education Union of Australia – Queensland and Northern Territory Branch	OSE .
ABN 74 662 601 045	DAN NEL CITE
346 Turbot Street SPRING HILL QLD 4000	PAUL NOEL GILES (Full Name) ASSISTANT SECRETARY/TREASURES (Position)
n the presence of -	(Position) 26. //. 202/ (Date) (Signature) (Signature) (Full Name) CLERTCAL OFFICER (Position) 26. //. 2021 (Date)
	{Dalei

SCHEDULE 1 - WAGES, SALARIES & ALLOWANCES TEACHING STAFF

Classification		2.5% Inc	rease		2.5% Increase				2.5% Increase				2.5% Increase				
	as at 1 January 2021			as at 1 January 2022				as at 1 January 2023				as at 1 January 2024					
	Fortnightly	Annual	Hourly	Casual	Fortnightly	Annual	Hourly	Casual	Fortnightly	Annual	Hourly	Casual	Fortnightly	Annual	Hourly	Casual	
Band 1												207					
Step 1	\$2,488.90	\$64,934	\$40.1435	\$50.1794	\$2,551.10	\$66,556	\$41.1468	\$51.4335	\$2,614.90	\$68,221	\$42.1758	\$52.7198	\$2,680.30	\$69,927	\$43.2306	\$54.0383	
Step 2	\$2,574.50	\$67,167	\$41.5242	\$51.9053	\$2,638.90	\$68,847	\$42.5629	\$53.2036	\$2,704.90	\$70,569	\$43.6274	\$54.5343	\$2,772.50	\$72,332	\$44.7177	\$55.8971	
Band 2																	
Step 1	\$2,822.20	\$73,629	\$45.5194	\$56.8993	\$2,892.80	\$75,471	\$46.6581	\$58.3226	\$2,965.10	\$77,357	\$47.8242	\$59.7803	\$3,039.20	\$79,290	\$49.0194	\$61.2743	
Step 2	\$2,960.10	\$77,227	\$47.7435	\$59.6794	\$3,034.10	\$79,157	\$48.9371	\$61.1714	\$3,110.00	\$81,138	\$50.1613	\$62.7016	\$3,187.80	\$83,167	\$51.4161	\$64.2701	
Step 3	\$3,100.70	\$80,895	\$50.0113	\$62.5141	\$3,178.20	\$82,917	\$51.2613	\$64.0766	\$3,257.70	\$84,991	\$52.5435	\$65.6794	\$3,339.10	\$87,115	\$53.8565	\$67.3206	
Step 4	\$3,246.60	\$84,701	\$52.3645	\$65.4556	\$3,327.80	\$86,820	\$53.6742	\$67.0928	\$3,411.00	\$88,990	\$55.0161	\$68.7701	\$3,496.30	\$91,216	\$56.3919	\$70.4899	
Band 3																	
Step 1	\$3,380.50	\$88,195	\$54.5242	\$68.1553	\$3,465.00	\$90,399	\$55.8871	\$69.8589	\$3,551.60	\$92,659	\$57.2839	\$71.6049	\$3,640.40	\$94,975	\$58.7161	\$73.3951	
Step 2	\$3,519.20	\$91,813	\$56.7613	\$70.9516	\$3,607.20	\$94,109	\$58.1806	\$72.7258	\$3,697.40	\$96,462	\$59.6355	\$74.5444	\$3,789.80	\$98,873	\$61.1258	\$76.4073	
Step 3	\$3,659.70	\$95,479	\$59.0274	\$73.7843	\$3,751.20	\$97,866	\$60.5032	\$75.6290	\$3,845.00	\$100,313	\$62.0161	\$77.5201	\$3,941.10	\$102,820	\$63.5661	\$79.4576	
Step 4	\$3,746.40	\$97,741	\$60.4258	\$75.5323	\$3,840.10	\$100,185	\$61.9371	\$77.4214	\$3,936.10	\$102,690	\$63.4855	\$79.3569	\$4,034.50	\$105,257	\$65.0726	\$81.3408	
Advanced	\$3,918.20	\$102,223	\$63.1968	\$78.9960	\$4,016.20	\$104,780	\$64.7774	\$80.9718	\$4,116.60	\$107,399	\$66.3968	\$82.9960	\$4,219.50	\$110,084	\$68.0565	\$85.0706	
Teacher	£4.007.00	#400 440	005.0040	000 0000	04.400.00	0400 700	207.0440	0040504	21.070.00								
Master Teacher 401	\$4,067.30	\$106,113	\$65.6016	\$82.0020	\$4,169.00	\$108,766	\$67.2419	\$84.0524	\$4,273.20	\$111,485	\$68.9226	\$86.1533	\$4,380.00	\$114,271	\$70.6452	\$88.3065	
Master Teacher 402					\$4,235.50	\$110,501	\$68.3145	\$85.3931	\$4,341.40	\$113,264	\$70.0226	\$87.5283	\$4,449.90	\$116,095	\$71.7726	\$89.7158	
Highly Accompl Tchr	\$4,389.50	\$114,519	\$70.7984	\$88.4980	\$4,499.20	\$117,381	\$72.5677	\$90.7096	\$4,611.70	\$120,316	\$74.3823	\$92.9779	\$4,727.00	\$123,324	\$76.2419	\$95.3024	
Lead Teacher	\$4,792.20	\$125,025	\$77.2935	\$96.6169	\$4,912.00	\$128,150	\$79.2258	\$99.0323	\$5,034.80	\$131,354	\$81.2065	\$101.5081	\$5,160.70	\$134,639	\$83.2371	\$104.0464	
Head of Faculty																	
Pay Point 1	4174.80	\$108,917	\$67.3355	\$84.1694	\$4,279.20	\$111,641	\$69.0194	\$86.2743	\$4,386.20	\$114,433	\$70.7452	\$88.4315	\$4,495.90	\$117,295	\$72.5145	\$90.6431	
Pay Point 2	4286.00	\$111,819	\$69.1290	\$86.4113	\$4,393.20	\$114,615	\$70.8581	\$88.5726	\$4,503.00	\$117,480	\$72.6290	\$90.7863	\$4,615.60	\$120,418	\$74.4452	\$93.0565	
Pay Point 3	4397.20	\$114,720	\$70.9226	\$88.6533	\$4,507.10	\$117,587	\$72.6952	\$90.8690	\$4,619.80	\$120,527	\$74.5129	\$93.1411	\$4,735.30	\$123,540	\$76.3758	\$95,4698	

SCHOOL OFFICERS

Classification	2.5% Inc	crease	2.5% lr	rcrease	2.5% Inc	crease	2.5% Increase as at 1 January 2024		
	as at 1 Janu	uary 2021	as at 1 Jar	nuary 2022	as at 1 Jani	Jary 2023			
	Fortnightly	Annual	Fortnightly	Annual	Fortnightly	Annual	Fortnightly	Annual	
Level 1	1		<u> </u>						
Step 1	\$1,668.30	\$43,525	\$1,710.00	\$44,613	\$1,752.80	\$45,729	\$1,796.60	\$46,872	
Step 2	\$1,705.80	\$44,503	\$1,748.40	\$45,614	\$1,792.10	\$46,755	\$1,836.90	\$47,923	
Step 3	\$1,744.00	\$45,500	\$1,787.60	\$46,637	\$1,832.30	\$47,803	\$1,878.10	\$48,998	
Step 4	\$1,782.20	\$46,496	\$1,826.80	\$47,660	\$1,872.50	\$48,852	\$1,919.30	\$50,073	
Level 2				<u>-</u>					
Step 1	\$1,819.50	\$47,469	\$1,865.00	\$48,656	\$1,911.60	\$49,872	\$1,959.40	\$51,119	
Step 2	\$1,876.70	\$48,962	\$1,923.60	\$50,185	\$1,971.70	\$51,440	\$2,021.00	\$52,726	
Step 3	\$1,895.40	\$49,450	\$1,942.80	\$50,686	\$1,991.40	\$51,954	\$2,041.20	\$53,253	
Level 3	<u> </u>						<u> </u>		
Step 1	\$1,895.40	\$49,450	\$1,942.80	\$50,686	\$1,991.40	\$51,954	\$2,041.20	\$53,253	
Step 2	\$1,933.50	\$50,444	\$1,981.80	\$51,704	\$2,031.30	\$52,995	\$2,082.10	\$54,320	
Step 3	\$2,028.20	\$52,914	\$2,078.90	\$54,237	\$2,130.90	\$55,594	\$2,184.20	\$56,984	
Step 4	\$2,085.30	\$54,404	\$2,137.40	\$55,763	\$2,190.80	\$57,156	\$2,245.60	\$58,586	
Level 4	 -					- 	·		
Step 1	\$2,123.10	\$55,390	\$2,176.20	\$56,775	\$2,230.60	\$58,195	\$2,286.40	\$59,650	
Step 2	\$2,179.80	\$56,869	\$2,234.30	\$58,291	\$2,290.20	\$59,750	\$2,347.50	\$61,245	
Step 3	\$2,236.70	\$58,354	\$2,292.60	\$59,812	\$2,349.90	\$61,307	\$2,408.60	\$62,839	
Level 5			-	<u> </u>	·				
Step 1	\$2,271.80	\$59,270	\$2,328.60	\$60,751	\$2,386.80	\$62,270	\$2,446.50	\$63,827	
Step 2	\$2,321.00	\$60,553	\$2,379.00	\$62,066	\$2,438.50	\$63,619	\$2,499.50	\$65,210	
Step 3	\$2,387.70	\$62,293	\$2,447.40	\$63,851	\$2,508.60	\$65,448	\$2,571.30	\$67,083	
Level 6		<u> </u>		~ _			L		
Step 1	\$2,482.30	\$64,761	\$2,544.40	\$66,382	\$2,608.00	\$68,041	\$2,673.20	\$69,742	
Step 2	\$2,599.10	\$67,809	\$2,664.10	\$69,504	\$2,730.70	\$71,242	\$2,799.00	\$73,024	
Step 3	\$2,718.40	\$70,921	\$2,786.40	\$72,695	\$2,856.10	\$74,514	\$2,927.50	\$76,376	
Step 4	\$2,824.70	\$73,694	\$2,895.30	\$75,536	\$2,967.70	\$77,425	\$3,041.90	\$79,361	
Step 5	\$2,887.80	\$75,341	\$2,960.00	\$77,224	\$3,034.00	\$79,155	\$3,109.90	\$81,135	
Level 7	 			L	<u> </u>		· ·		
Step 1	\$3,108.60	\$81,101	\$3,186.30	\$83,128	\$3,266.00	\$85,208	\$3,347.70	\$87,339	
Step 2	\$3,165.30	\$82,580	\$3,244.40	\$84,644	\$3,325.50	\$86,760	\$3,408.60	\$88,928	
Step 3	\$3,222.50	\$84,073	\$3,303.10	\$86,175	\$3,385.70	\$88,330	\$3,470.30	\$90,538	
Step 4	\$3,298.30	\$86,050	\$3,380.80	\$88,203	\$3,465.30	\$90,407	\$3,551.90	\$92,666	
Step 5	\$3,373.90	\$88,023	\$3,458.20	\$90,222	\$3,544.70	\$92,479	\$3,633.30	\$94,790	

TAFE TUTORS/TRAINERS

Classification	Effective as at 1 Jan 2021			2.5% Increa	ise from 1 Jan	2022	2.5% Increa	se from 1 Jan	2.5% Increase from 1 Jan 2024			
	Fortnightly	Annually	Hourly	Fortnightly	Annually	Hourly	Fortnightly	Annually	Hourly	Fortnightly	Annually	Hourly
Foundation Educator	2,090.10	54,529	27.50	2,142.40	55,894	28.19	2,196.00	57,292	28.89	2,250.90	58,724	29.62
Tutor 1	2,148.50	56,053	28.27	2,202.20	57,454	28.98	2,257.30	58,891	29.70	2,313.70	60,363	30.44
Tutor 2	2,299.80	60,000	30.26	2,357.30	61,500	31.02	2,416.20	63,037	31.79	2,476.60	64,613	32.59
Tutor 3	2,320.00	60,527	30.53	2,378.00	62,040	31.29	2,437.50	63,593	32.07	2,498.40	65,181	32.87
Senior Tutor 1	2,602.80	67,905	34.25	2,667.90	69,604	35.10	2,734.60	71,344	35.98	2,803.00	73,128	36.88
Senior Tutor 2	2,711.20	70,733	35.67	2,779.00	72,502	36.57	2,848.50	74,315	37.48	2,919.70	76,173	38.42
											<u></u>	
Trainer 1	2,955.30	77,102	38.89	3,029.20	79,030	39.86	3,104.90	81,005	40.85	3,182.50	83,029	41.88
Trainer 2	3,086.80	80,532	40.62	3,164.00	82,546	41.63	3,243.10	84,610	42.67	3,324.20	86,726	43.74
Trainer 3	3,225.00	84,138	42.43	3,305.60	86,241	43.49	3,388.20	88,396	44.58	3,472.90	90,605	45.70
Trainer 4	3,356.70	87,574	44.17	3,440.60	89,763	45.27	3,526.60	92,006	46.40	3,614.80	94,307	47.56
Senior Trainer 1	3,491.80	91,098	45.94	3,579.10	93,376	47.09	3,668.60	95,711	48.27	3,760.30	98,103	49.48
Senior Trainer 2	3,628.40	94,662	47.74	3,719.10	97,029	48.94	3,812.10	99,455	50.16	3,907.40	101,941	51.41
Senior Trainer 3							3,907.20	101,936	51.41	4,004.90	104,485	52.70

SOCIAL AND COMMUNITY SERVICES AWARD

Classification	Effective	as at 1 Jul 2	2020	2.5% Increa	ase from 1 J	ul 2021	2.5% Increa	ase from 1 J	lul 2022	2.5% Increa	ase from 1 J	ul 2023	2.5% Increa	ase from 1 J	lul 2024
	Fortnightly	Annually	Hourly	Fortnightly	Annually	Hourly	Fortnightly	Annually	Hourly	Fortnightly	Annually	Hourly	Fortnightly	Annually	Hourly
Level 3															
Pay Point 1	2,413.16	62,958	31.75	2,473.50	64,532	32.55	2,535.30	66,144	33.36	2,598.70	67,798	34.19	2,663.70	69,494	35.05
Pay Point 2	2,482.46	64,766	32.66	2,544.50	66,384	33.48	2,608.10	68,043	34.32	2,673.30	69,744	35.18	2,740.10	71,487	36.05
Pay Point 3	2,535.38	66,146	33.36	2,598.80	67,801	34.19	2,663.80	69,497	35.05	2,730.40	71,234	35.93	2,798.70	73,016	36.83
Pay Point 4	2,587.28	67,500	34.04	2,652.00	69,189	34.89	2,718.30	70,918	35.77	2,786.30	72,693	36.66	2,856.00	74,511	37.58
Level 4															
Pay Point 1	2,783.08	72,608	36.62	2,852.70	74,425	37.54	2,924.00	76,285	38.47	2,997.10	78,192	39.44	3,072.00	80,146	40.42
Pay Point 2	2,855.96	74,510	37.58	2,927.40	76,374	38.52	3,000.60	78,283	39.48	3,075.60	80,240	40.47	3,152.50	82,246	41.48
Pay Point 3	2,929.34	76,424	38.54	3,002.60	78,336	39.51	3,077.70	80,295	40.50	3,154.60	82,301	41.51	3,233.50	84,360	42.55
Pay Point 4	2,994.56	78,126	39.40	3,069.40	80,078	40.39	3,146.10	82,079	41.40	3,224.80	84,133	42.43	3,305.40	86,235	43.49
Level 5															
Pay Point 1	3,183.88	83,065	41.89	3,263.50	85,142	42.94	3,345.10	87,271	44.01	3,428.70	89,452	45.11	3,514.40	91,688	46.24
Pay Point 2	3,252.10	84,845	42.79	3,333.40	86,966	43.86	3,416.70	89,139	44.96	3,502.10	91,367	46.08	3,589.70	93,653	47.23
Pay Point 3	3,328.00	86,825	43.79	3,411.20	88,996	44.88	3,496.50	91,221	46.01	3,583.90	93,501	47.16	3,673.50	95,839	48.34
Level 6															
Pay Point 1	3,478.44	90,750	45.77	3,565.40	93,019	46.91	3,654.50	95,343	48.09	3,745.90	97,728	49.29	3,839.50	100,170	50.52
Pay Point 2	3,555.16	92,751	46.78	3,644.00	95,069	47.95	3,735.10	97,446	49.15	3,828.50	99,883	50.38	3,924.20	102,379	51.63
Pay Point 3	3,632.16	94,760	47.79	3,723.00	97,130	48.99	3,816.10	99,559	50.21	3,911.50	102,048	51.47	4,009.30	104,600	52.75
Level 7				A											
Pay Point 1	3,762.14	98,151	49.50	3,856.20	100,605	50.74	3,952.60	103,120	52.01	4,051.40	105,698	53.31	4,152.70	108,341	54.64

SERVICES STAFF

Classification	Effecti	ive as at 1s	st July 20	21	2.5% Increase from 1st July 2022		2.5% Increase from 1st July 2023			2.5% Inc	2.5% Increase from 1st July 2024					
Level	Fortnightly	Annual	P/T	Casual	Fortnightly	Annual	P/T	Casual	Fortnightly	Annual	P/T	Casual	Fortnightly	Annual	P/T	Casual
1.1	\$1,589.60	\$41,471	\$20.92	\$26.14	\$1,629.30	\$42,507	\$21.44	\$26.80	\$1,670.00	\$43,569	\$21.97	\$27.47	\$1,711.80	\$44,660	\$22.52	\$28.16
1.2	\$1,650.40	\$43,058	\$21.72	\$27.14	\$1,691.70	\$44,135	\$22.26	\$27.82	\$1,734.00	\$45,239	\$22.82	\$28.52	\$1,777.40	\$46,371	\$23.39	\$29.23
1.3	\$1,711.00	\$44,639	\$22.51	\$28.14	\$1,753.80	\$45,755	\$23.08	\$28.85	\$1,797.60	\$46,898	\$23.65	\$29.57	\$1,842.50	\$48,069	\$24.24	\$30.31
2.1	\$1,724.20	\$44,983	\$22.69	\$28.36	\$1,767.30	\$46,108	\$23.25	\$29.07	\$1,811.50	\$47,261	\$23.84	\$29.79	\$1,856.80	\$48,443	\$24.43	\$30.54
2.2	\$1,778.00	\$46,387	\$23.39	\$29.24	\$1,822.50	\$47,548	\$23.98	\$29.97	\$1,868.10	\$48,737	\$24.58	\$30.72	\$1,914.80	\$49,956	\$25.19	\$31.49
3.1	\$1,800.20	\$46,966	\$23.69	\$29.61	\$1,845.20	\$48,140	\$24.28	\$30.35	\$1,891.30	\$49,343	\$24.89	\$31.11	\$1,938.60	\$50,577	\$25.51	\$31.89
3.2	\$1,832.40	\$47,806	\$24.11	\$30.14	\$1,878.20	\$49,001	\$24.71	\$30.89	\$1,925.20	\$50,227	\$25.33	\$31.66	\$1,973.30	\$51,482	\$25.96	\$32.46
4 .1	\$1,900.20	\$49,575	\$25.00	\$31.25	\$1,947.70	\$50,814	\$25.63	\$32.03	\$1,996.40	\$52,085	\$26.27	\$32.84	\$2,046.30	\$53,386	\$26.93	\$33.66
4.2	\$1,995.40	\$52,059	\$26.26	\$32.82	\$2,045.30	\$53,360	\$26.91	\$33.64	\$2,096.40	\$54,694	\$27.58	\$34.48	\$2,148.80	\$56,061	\$28.27	\$35.34
5.1	\$2,060.20	\$53,749	\$27.11	\$33.88	\$2,111.70	\$55,093	\$27.79	\$34.73	\$2,164.50	\$56,470	\$28.48	\$35.60	\$2,218.60	\$57,882	\$29.19	\$36.49
5.2	\$2,158.80	\$56,321	\$28.41	\$35.51	\$2,212.80	\$57,730	\$29.12	\$36.39	\$2,268.10	\$59,173	\$29.84	\$37.30	\$2,324.80	\$60,652	\$30.59	\$38.24
6.1	\$2,236.40	\$58,346	\$29.43	\$36.78	\$2,292.30	\$59,804	\$30.16	\$37.70	\$2,349.60	\$61,299	\$30.92	\$38.65	\$2,408.30	\$62,831	\$31.69	\$39.61
6.2	\$2,388.60	\$62,317	\$31.43	\$39.29	\$2,448.30	\$63,874	\$32.21	\$40.27	\$2,509.50	\$65,471	\$33.02	\$41.28	\$2,572.20	\$67,107	\$33.85	\$42.31
7.1	\$2,458.60	\$64,143	\$32.35	\$40.44	\$2,520.10	\$65,748	\$33.16	\$41.45	\$2,583.10	\$67,391	\$33.99	\$42.48	\$2,647.70	\$69,077	\$34.84	\$43.55
7.2	\$2,537.40	\$66,199	\$33.39	\$41.73	\$2,600.80	\$67,853	\$34.22	\$42.78	\$2,665.80	\$69,549	\$35.08	\$43.85	\$2,732.40	\$71,286	\$35.95	\$44.94
7.3	\$2,615.80	\$68,244	\$34.42	\$43.02	\$2,681.20	\$69,951	\$35.28	\$44.10	\$2,748.20	\$71,699	\$36.16	\$45.20	\$2,816.90	\$73,491	\$37.06	\$46.33
8	\$2,848.80	\$74,323	\$37.48	\$46.86	\$2,920.00	\$76,181	\$38.42	\$48.03	\$2,993.00	\$78,085	\$39.38	\$49.23	\$3,067.80	\$80,037	\$40.37	\$50.46

SCHEDULE 2 - STATEMENT OF FAITH

The Company has the following doctrinal statement which all members and office bearers must subscribe:

- (a) We believe that the Bible is the inspired and infallible Word of God:
- (b) We believe in one God, eternally existent in three persons. Father, Son and Holy Spirit;
- (c) We believe in the Lord Jesus Christ, the second person of the triune Godhead, who was and is the eternal Son of God, that he became incarnate by the Holy Spirit and was born of the virgin Mary.
- (d) We believe in his sinless life, miraculous ministry, substitutionary, atoning death, bodily resurrection, glorious ascension, and abiding intercession.
- (e) We believe that he will come again to this earth at the end of the age to judge the living and the dead and to reign as King of kings and Lord of lords;
- (f) We believe that salvation is by grace through faith in the Lord Jesus Christ; that it is a work of the Holy Spirit whereby the repentant and believing sinner is born again and becomes a new creation in Christ Jesus;
- (g) We believe in the sanctifying power of the Holy Spirit which enables a believer to live a holy and God-glorifying life.

SCHEDULE 3 - CONDITIONS SPECIFIC TO SCHOOL OFFICERS

1. Employment Relationship

1.1 Contract of Employment

- 1.1.1 Each employee other than a casual employee shall be advised in writing at point of engagement and at other times when varied in accordance with this Schedule, the following:
 - (a) The nature of engagement as either full-time or such other category as provided in clause 1.1.2.
 - (b) If not full-time, the weeks the employee is to be employed.
 - (c) The days of the week the employee is to be employed.
 - (d) The normal starting and finishing time for each day's employment.
 - (e) The duration of the engagement in respect of employment for a fixed period.
- 1.1.2 Employment categories are:
 - (a) full-time which means an employee engaged to work 38 ordinary hours per week on the basis of 52 weeks per annum;
 - (b) part-time;
 - (c) casual; or
 - (d) term-time employee; or
 - (e) fixed period employee.
- 1.1.3 Subject to clauses 7.2 and 8.2 of this Agreement, an employer may vary the terms of engagement of any employee other than a casual employee by providing 2 weeks' notice of such changes unless it is mutually agreed between the employer and employee for a shorter period of time.

1.2 Part-time Employment

- 1.2.1 A part-time employee is an employee who:
 - (a) is employed for less than 38 ordinary hours per week on the basis of 52 weeks per annum; and
 - (b) has reasonably predictable hours of work; and
 - (c) receives, on a pro rata basis, equivalent pay and conditions to those of full-time employees covered by this Schedule.
- 1.2.2 At the time of engagement, the employer and the employee will agree in writing on the pattern of work required, including specifying the number of ordinary hours per week, the days on which the work is to be performed and the normal daily starting and finishing times.

- 1.2.3 Any variation to the work pattern, including the normal starting and finishing times prescribed in clause 6.1.3, will be in accordance with methods of altering the ordinary hours of work for full-time employees.
- 1.2.4 Subject to clause 1.1.3 the agreed number of ordinary hours per week may be varied by mutual agreement. Any such agreed variation to the number of weekly hours of work will be recorded in writing.
- 1.2.5 All time worked outside the spread of ordinary working hours and the employees normal starting and finishing times of ordinary hours will be overtime and paid for at the rates prescribed in clause 5.5 Overtime, of this Schedule.
- 1.2.6 A part-time employee employed under the provisions of clause 1.2 must be paid for ordinary hours worked at the rate of 1/38th of the weekly rate prescribed for the class of work performed.
- 1.2.7 Where a public holiday falls on a day upon which an employee is normally employed, that employee shall be paid the appropriate rate for the number of hours normally worked on that day.
- 1.2.8 Where an employee and their employer agree in writing, part-time employment may be converted to full-time, and vice-versa. If such an employee transfers from full-time to part-time (or vice-versa), all accrued Agreement and legislative entitlements shall be maintained. Following transfer to part-time employment accrual will occur in accordance with the provisions relevant to part-time employment.

1.3 Casual Employment

- 1.3.1 Casual employee means an employee engaged and paid as such and who is employed by the hour for a maximum period of 20 working days on any one engagement.
- 1.3.2 A casual employee shall be paid an hourly rate equal to 1/38th of the weekly rate for a full-time employee plus 25% with a minimum payment of 2 hours for each day's engagement.

1.4 Employment of term-time and fixed periodemployees

- 1.4.1 Term-time or fixed period employees shall be entitled to be paid an hourly rate for ordinary hours worked equal to the appropriate weekly full-time rate divided by 38.
- 1.4.2 Employees under clause 1.4 shall be entitled to receive pro rata entitlements to annual leave and personal leave in accordance with clauses 5.1.1 and 5.6 of this Agreement.
- 1.4.3 Where a public holiday falls on a day upon which an employee is normally employed, that employee shall be paid the appropriate rate for the number of hours normally worked on that day.
- 1.4.4 "Term-time employee" is a continuing employee engaged to work:
 - (a) 38 ordinary hours per week but less than 52 weeks per annum; or
 - (b) Less than 38 ordinary hours per week and less than 52 weeks per annum.
- 1.4.5 "Fixed period employee" is one engaged to work 38 ordinary hours or less per week with a specified commencement and cessation date:

Provided that a fixed period employee may be engaged for a period of up to 52 weeks in

respect of any one engagement.

2. Classification

- 2.1 An employer shall determine the classification of a position through the following process.
 - (a) An analysis is to be undertaken to establish the skills and responsibilities required for each identified position and a position description written for each position.
 - (b) Each position is classified by reference to the classification criteria set out in clause 2.4 using the position description developed in accordance with clause 2.1.1 (a).
 - (c) Employees are appointed to a position at the appropriate level within the structure and to a step in the level according to experience based on years of service.
- 2.2 If at any time an employee or an employer considers that the skills and responsibilities as required by the employer for a position have altered or do not reflect the classification determined, a review of the classification applicable to the position is to be undertaken in accordance with clause 2.1 and an appropriate classification determined. However, except in exceptional circumstances such as a change in the skill and/or responsibility required, or a change in the conditions under which the work is performed, no employee shall be permitted to seek a reclassification of their position on more than one occasion in a 12-month period.
- 2.3 Where a disagreement arises as to the outcome of a classification review, it will be dealt with under the grievance procedure contained in clause 7.1 of this Agreement. At any meeting specified in clause 7.1 of this Agreement the person who made the decision about the classification review shall, wherever possible participate.

2.4 Classification Criteria

- 2.4.1 Classification criteria are guidelines to determine the appropriate classification level and consist of characteristics and typical duties and skills.
- 2.4.2 The characteristics are the principal guide to classification as they are designed to indicate the level of basic knowledge, comprehension of issues, problem and procedures required, the level of autonomy, accountability supervision/training involved with the position. The characteristics of a level must be read as a whole to gain an understanding of the position and the performance requirements. Isolated characteristics should not be used to justify the classification of a position.
- 2.4.3 The typical duties/skills are a non-exhaustive list of duties/skills that may be comprehended within the particular level. They reflect the competencies of a particular level. They are an indicative guide only and at any particular level employees may be expected to undertake duties of any level lower than their own. Employees at any particular level may perform/utilise one such duty/skill or many of them depending on the particular work allocated to them. Typical duties/skills should not be used as the primary determinant in classifying employees but may be useful if the characteristics of a level cannot be easily applied in an individual instance.
- 2.5 Some of the characteristics have been included in the typical skills/duties at each level. Where there is inconsistency between the characteristics and the typical skills/duties, the characteristics will prevail over the typical skills/duties.
 - (a) The key issue to be looked at in properly classifying an employee is the level of

- initiative, responsibility/accountability, competency and skill that an employee is required to exercise in performing the employee's work within the parameters of the characteristics, read as a whole, of the position.
- (b) It should be noted that some typical duties/skills appear at one level only while others appear at more than one level. Because of this, the classification or reclassification of a position needs to be done by reference to the specific characteristics of the level. As an example, because an employee may be utilising a skill comprehended at a higher level than that to which the employee has been appointed, the employee assumes the level of initiative, accountability/responsibility, skill and competency envisaged by the characteristics of the higher level.
- 2.6 Positions of all school officers will be classified according to this Schedule.

2.7 Reclassification Process

- 2.7.1 A school officer may request a reclassification of their position. Such a request may be made either in relation to the classification level of an existing position or where the classification level of the position has been changed. Except in exceptional circumstances no employee shall be permitted to seek a reclassification of their position on more than one occasion in a 12 month period.
- 2.7.2 The employee shall make any such Request for Reclassification, in writing, to the employer.
- 2.7.3 The employer shall consider the Request for Reclassification and notify the employee in writing of the decision regarding the employee's request.
- 2.7.4 If after receiving the employer's notification, the employee believes that their position has not been classified at the correct level, the employee may apply for a review of that decision. In this case the employee shall make written application for a Review of Classification to the employer.
- 2.7.5 Where the employer receives a Review of Classification application, the employer shall advise the Independent Education Union of Australia Queensland and Northern Territory (IEUA-QNT) in writing that an application has been received.
- 2.7.6 A joint review of the classification by IEUA-QNT and the employer shall then take place. The persons representing IEUA-QNT and the employer will, wherever practicable, have substantial experience in the classification of school officers. Further, wherever practicable, the person who made the original decision in relation to the classification will also be involved.
- 2.7.7 The joint review will seek to reach a consensus position. The school officer will be advised in writing of the outcome of this review.
- 2.7.8 If an agreed outcome cannot be reached between the employer and IEUA-QNT then the employee may refer the matter to The Fair Work Commission in the terms of the Grievance Procedure.
- 2.7.9 The provisions of this clause 2.7.2 will be in addition to the provisions of clause 2 of this Schedule.

3. Incremental Advancement

3.1 Each level of the structure has varying pay steps which provide for yearly service increments within a level. Such increments are payable subject to satisfactory performance but will not be unreasonably withheld by the employer without due process.

- 3.2 For the purposes of establishing the entitlement of an employee to a yearly pay increment a year's service shall constitute 1976 hours of duty.
- 3.3 Progression from one level to a higher level is either by appointment to such higher level as a result of vacancy at that level or the employer requiring an employee to perform at a higher level in accordance with the classification criteria set out in clause 2.4.
- 3.4 An employee may be appointed to a higher level without having progressed through all pay steps within a lower level.

4. Recognition of Service - School Officers

This clause is to be used to determine the incremental step in the classification level in accordance with clauses 1 and 2 of this Schedule.

- 4.1 Recognition of years of service for incremental purposes will include all previous service as a school officer within non-governmental schools at or above the classification level of the position to which the employee is appointed.
- 4.2 An employee may make application for recognition of previous service other than as a school officer in a non-governmental school. The recognition of this other service will be based upon demonstrated relevance to the work of the position to which the employee has been appointed.
- 4.3 The provision of documentary evidence of previous employment as a school officer will be the responsibility of the employee.
- 4.4 Only service in the ten (10) years prior to the date of application will be considered for recognition.

4.5 Timely notification of previous relevant service

- 4.5.1 Employees will provide the employer with timely written advice along with supporting documentary evidence of relevant employment service which would be relevant to determining the employee's rate of pay.
- 4.5.2 If an employee does provide supporting documentary evidence regarding previous relevant service within six (6) months of commencement of service with the employer, payment for the adjustment will be effective from the date of commencement with the employer.
- 4.5.3 If an employee does not provide supporting documentary evidence regarding relevant service within six (6) months of commencement then any payment relating to that service will be applied from the date of receipt of notification from the employee.
- 4.5.4 Supporting documentary evidence of previous relevant service should include statements of service. A Statement of Service from each previous employer(s) should be provided. A Statement of Service is normally the appropriate proof of prior service and satisfies the criteria at clause 4.6 of this Schedule (below).
- 4.5.5 A statutory declaration may be submitted in lieu of a Statement of Service in instances where an individual is unable to obtain the necessary documentation from a previous employer. For instance, a school and its records may no longer exist or exist in an area of civil disturbance or natural disaster. A statutory declaration may be submitted only after the employer is satisfied that the employee has exhausted all reasonable avenues to obtain a Statement of Service. Where a statutory declaration is accepted it should satisfy the criteria

at clause 4.6 of this Schedule (below). The circumstances preventing the employee from obtaining the necessary documentation should be noted and retained with the application.

4.6 Statement of Service

The Statement of Service should:

- (a) be an original or certified copy of an original document;
- (b) be provided on the official letterhead of the authority responsible for the school or institution and it must be signed and dated by the employer or an authorised person;
- (c) specify the position held;
- (d) specify the period of employment;
- specify the exact nature of the employment e.g. full-time, part-time, casual, etc. and provide a detailed description of the relevant duties performed;
- (f) detail whether the employment was part-time and if so the number of hours worked per week or their full-time equivalent, or the total number of hours paid for the total period of employment;
- (g) detail whether the employment was casual and if so the total number of hours paid for the total period of employment;
- (h) indicate any periods of unpaid leave that were taken; and
- (i) specify whether any leave without pay was taken and the period when this leave without pay occurred. If no periods of leave without pay were taken, the statement must show that 'no leave without pay was taken'.

4.7 Duty to Notify

All new employees will be given at the time of appointment a document detailing the requirements with regards to timely notification of gaining qualifications and timely notification of previous relevant service.

5. Hours of Work

- 5.1 Subject to the provisions of this clause the ordinary hours of work for a school officer shall not exceed 38 hours per week.
- 5.2 Such ordinary hours of work shall be worked continuously (except for meal breaks) between 7 am and 6 pm on Mondays to Fridays inclusive.
- 5.3 The normal starting and finishing times of ordinary hours shall be established at the point of engagement. The normal starting and finishing times can only be varied:
 - (a) By the employer giving two weeks' notice of the change, or
 - (b) Where the employee agrees to the change.
- 5.4 Notwithstanding clause 5.3 (a) above, an employee's roster cannot be changed where it would result in the employee becoming ineligible for payment on a public holiday, in

circumstances where the employee would otherwise have been rostered to work on that day other than at the election of the employee.

5.5 Overtime

5.5.1 Except as provided in clause 4.1 of this Schedule, all time worked outside of the hours prescribed in clause 5.2 and the normal starting and finishing times shall be paid for at the rate of time and one-half for the first 3 hours and double time thereafter.

5.5.2 Time in Lieu

- (a) Where the employer requires an employee to work either before the normal starting times or after the normal finishing times contained in clause 5.2 of this Schedule, up to a maximum of 2 hours on any one day the employee may be granted time in lieu for the equivalent hours worked to be taken at a mutually convenient time or receive payment at the rate of time and one half for the time worked.
- (b) Where the employer requires the employee to work more than 2 hours before the normal starting times or more than 2 hours beyond the normal finishing times on any one day, the employee may receive payment at the rate of time and a half for the first hour and double time thereafter for time worked beyond the first 2 hours or receive payment in accordance with clause 5.5.2 of this Schedule for the full period so worked.
- (c) Any accrued time in lieu not taken within 6 months of its accrual shall be paid at the rate of time and one-half.

6. Allowances

6.1 Qualifications/ First Aid Allowance

- (a) A school officer who has been on Level 2 Step 3 for twelve months (or 1976 hours for other than full-time employees), and who holds a Certificate Level (III) qualification (or higher) which is relevant to their work, and who holds a current First Aid certificate will receive the Qualification Allowance identified in Wages Schedule.
- (b) A school officer who has been on Level 3 Step 4 for twelve months (or 1976 hours for other relevant to their work, and who holds a current First Aid certificate will receive the Qualification Allowance identified in Schedule 1.
- (c) A school officer who has been on Level 4 Step 3 for twelve months (or 1976 hours for other than full-time employees), and who holds a Diploma or Associate Diploma level qualification (or higher) which is relevant to their work, and who holds a current First Aid certificate will receive the Qualification Allowance identified in Schedule 1.
- (d) A school officer who has been on Level 5 Step 3 for twelve months (or 1976 hours for other than full-time employees), and who holds a Degree level qualification (or higher) which is relevant to their work, and who holds a current First Aid certificate will receive the Qualification Allowance identified in Schedule 1.
- (e) A school officer will apply for the payment of the allowance prescribed in 6.1.2 (a), (b), (c) or (d). Where the employee applies for the allowance within six (6) months of the qualification being obtained the allowance will be paid as from the date that the qualification is obtained. In other circumstances the school officer will be paid the allowance from the date of making the application.

- (f) Where a Level 2, 3, 4 or 5 school officer is receiving the Qualification Allowance in accordance with this clause, the employer will pay any fees associated with obtaining and maintaining the First Aid certificate. The employee will contribute the time required to maintain the First Aid certificate.
- (g) The employer will reimburse enrolment fees associated with obtaining a First Aid certificate which are incurred within the twelve (12) month period immediately prior to the school officer accessing the qualification allowance in clause 6.1.2 (a), (b), (c) or (d) as appropriate. The employee will contribute the time required to obtain the first aid certificate.
- (h) A school officer may receive either the Qualifications Allowance (as in clauses 6.1.2 (a), (b), (c) or (d) or the First Aid Allowance (as in clause 6.2) but not more than one of these allowances.

6.2 School officer - First Aid Allowance

A school officer who holds a current First Aid certificate and who is appointed by the employer to be a First Aid Officer will receive a First Aid Allowance at the rate of \$18.20 per week.

6.3 Higher Duties Allowance

- (a) The employer may require a school officer to act temporarily in a position which has a classification higher than the employee's current classification level.
- (b) Where the employee is appointed to act for more than one week, the employee will be paid at the first incremental point of the higher classification level for the whole period of appointment.

7. Annualisation of Salary-Term Time Employees

- 7.1 Term Time School Officers who work less than a full year may agree to have their salary annualised over the full year. Such an agreement will be:
 - (a) recorded in writing; and
 - (b) kept as part of the times and wages record.
- 7.2 Hours in excess of the Term Time School Officers agreed normal working hours will be paid within the next pay period.
- 7.3 Where there are more than short term changes to the standard roster of hours a new calculation of annualisation of wages will occur.

8. Rest Pauses and Meal Breaks

8.1 Rest pauses

- 8.1.1 Full-time employees shall receive a paid rest pause of 10 minutes' duration in the first half and the second half of each day worked.
- 8.1.2 Employees other than full-time are entitled to a rest break of 10 minutes for each period of 3 hours worked (excluding meal breaks) with a maximum of 2 rest breaks per shift. Such rest pauses shall be taken at such times as will not interfere with continuity of work where continuity is necessary.

8.1.3 Notwithstanding the foregoing, where the employer and the employees agree the rest pauses may be combined.

8.2 Meal Break

An employee shall be entitled to an unpaid meal break of not less than half an hour and not more than one hour per working day.

1. CHARACTERISTICS AND QUALIFICATION

Competency of employees

Level 2	Level 3	Level 4	Level 5	Level 6	Level 7
Competency at this level involves application of knowledge and skills to a range of tasks and roles	Competency at this level involves application of knowledge with depth in some areas and a broad range of skills		Competency at this level involves self-directed application of knowledge with substantial depth in some areas	Competency at this level involves self-directed development and application of professional knowledge with substantial depth in some areas	Competency at this level involves the use of initiative in self directed development and application of expert knowledge with extensive recognised expertise in some areas
There is a defined range of contexts where the choice of actions required is clear	There is a range of roles and tasks in a variety of contexts	There is a wide variety of tasks and roles in a variety of contexts.	A range of technical and/or other skills are applied to roles and functions in both varied and highly specific contexts.	A broad range of professional skills are applied to roles and functions in both varied and highly specific contexts.	A breadth and depth of professional skills are applied to roles and functions in both varied and highly specific contexts.
There is limited complexity of choice	There is some complexity in the extent and choice of actions required	There is complexity in the ranges and choice of actions required		A proportion of competencies involve complex, specialized or professional functions.	A high proportion of competencies involve significant scope and/or complex, specialized or professional functions.
Competencies are normally checked within well established routines, methods and procedures	Competencies are normally used within routines, methods and procedures	Competencies are normally used within a variety of routines, methods and procedures	Competencies are normally used independently and both routinely and non routinely.	Competencies are used independently and are substantially non-routine with initiative being exercised in the application of professional practices	Duties of an innovative and/or critical nature are undertaken without professional direction and initiative is exercised in the application of professional practices
Limited discretion and judgement about possible actions is involved	Some discretion and judgement are involved in selection of equipment, work organisation, services, actions and achieving outcomes within time constraints.	Discretion and judgement are required for self and/or others in planning, selection of equipment, work organisation, services actions and achieving outcomes within time constraints.	Discretion and judgement are required in planning and selecting appropriate equipment, service techniques and work organisation for self and/or others.	Significant discretion and judgement are required in planning, design, professional, technical or supervisory functions related to services, operations or processes for self and/or others.	Significant discretion and independent judgement are required within constraints set by management.

Supervision of Employees Work

Level 2	Level 3	Level 4	Level5	Level 6	Level 7
Works under direct and/or routine supervision depending on function.	Works under limited supervision.	Work is carried out under general supervision.	Works under general supervision and/or broad guidance depending on function.	Works under broad guidance.	Work is usually performed under general guidance with limited or no professional supervision.
Work is intermittently checked.	Work may be checked in relation to overall progress			Work is usually measured in terms of the achievement of staled objectives to agreed standards.	The general quality of work is monitored by school management and is subject to stated objectives and professional standards.
May take the form of general guidance where working in teams is involved.	May take the form of broad guidance.	Progress and outcomes sought are under general guidance.		May be less direct than at lower levels and usually be related to task methodology and work practices.	
May involve detailed instructions in some situations;	May involve a level of autonomy when working in teams.			May involve a level of autonomy in accordance with a broad plan or budget strategy.	

Supervision of Others

Level 2	Level 3	Level 4	Level 5	Level 6	Level 7
An employee at this level will have no supervisory responsibilities	Peer assistance may be provided to others.	The work of others may be supervised			
	An employee may have limited responsibility for guidance of the work of others	Responsibility for the work and organisation of others in limited areas.	Responsibility for the planning and management of the work of others may be involved	Responsibility for the supervision and monitoring of the work of others and of workflow in the area of responsibility may be involved.	Responsibility for the setting and achieving of objectives by a work section and its staff may be involved.
An experienced employee may assist others by providing peer support in the completion of routine tasks	Learn co-ordination may be required	Teams may be guided or facilitated.		Leadership and development of teams and responsibility for outcomes may be required.	
		Training of subordinate staff may be required.	Supervision and training of lower level staff may be involved.		Responsibility for assessment, training and development and performance counselling of staff may be required.

Levels1to7QualificationsMatrix Level 1 Level 2 Level 3 Level 4 Level 5 Level 6 Level 7 Junior Certificate is the Junior Certificate is the Tertiary qualifications Tertiary qualifications Tertiary qualifications Formal qualifications at Relevant formal at Associate minimum formal minimum formal at Certificate level or at Certificate level or qualifications at degree degree level are equivalent equivalent Diploma/Diploma level qualification. No qualification. No level are required. required, along with experience is required. qualifications relevant qualifications relevant or equivalent experience is required. relevant post graduate qualifications relevant to the position may be to the position may be qualifications or to the position may be required or such required or such extensive and relevant knowledge, knowledge, required by the experience as required qualifications and qualifications and employer or by the employer to experience that are experience that are knowledge, reflect higher levels of deemed by the deemed by the qualifications and professional outcomes. employer as employer as experience that are necessary to necessary to determined by the successfully carry out successfully carry out employer as necessary the duties of the to successfully carry the duties of the out the duties of the position. position. position.

2. Typical Duties/Skills Employee Assisting Student Learning

Level 2	Level 3	Level 4	Level 5	Level 6	Level 7
2.1.1 Assist developmentally appropriate student learning, either individually or in groups, under the direct supervision of an academic staff member where limited discretion and judgement are involved. 2.1.2 Under direct supervision of a higher level officer or members of the academic staff prepare and clear away materials for display/use in classrooms or libraries. 2.1.3 Perform within wellestablished routines tasks associated with the mass production of printed material including collating, stapling, binding, folding, cutting, etc 2.1.4 Under direct and /or routine supervision, perform tasks of limited complexity, associated with classroom learning experiences, such as assisting teachers in preparing, implementing and supervising learning programs. 2.1.5 Support students in relation to their physical needs.	3.1.1 Assist developmentally appropriate student learning, either individually or in groups, where some discretion and judgement are involved in evaluating and assessing (under the supervision of an academic staff member(s)) the learning needs of students. 3.1.2 Within routines, methods and procedures carry out liaison between the school, the student and the student's family where some discretion and judgement are involved. 3.1.3 Support students in relation to their physical needs where some discretion and judgement are involved.	4.1.1 Assist developmentally appropriate student learning, either individually oringroups, under the general supervision of an academic staff member(s). Employees at this level are required to exercise discretion and judgement to modify education programs to meet the learning needs of specific students. 4.1.2 Carry out liaison between the school, the student and the student and the student are required in relation to planning, actions and achieving outcomes. 4.1.3 Within a variety of routines, methods and procedures provide significant assistance in the enrolment, family liaison and placement of overseas students.	5.1.1 Apply a range of technical and other skills involving the self-directed application of knowledge gained through formal studies/qualifications applicable to this level or knowledge and experience that are determined by the employer as necessary to successfully carry out the duties of the position. This may include: developing the framework for and providing the instruction to students (within a structured learning environment) under the general supervision of an academic staff member(s); providing pastoral ministry and support for students. 5.1.2 Under broad guidance, supervise the operations of the school's processes and activities in relation to overseas students. This may include: enrolment; family liaison; and placement.	6.1.1 Apply a range of professional knowledge gained through successful completion of an appropriate undergraduate degree. This may include: the gathering, analysis and interpretation of data; or preparation of reports and the consequent giving of advice to other professional staff to assist student learning; or providing pastoral ministry; or providing counselling and/or guidance support for students. 6.1.2 Provide professional advice to staff and students in the officer's area of expertise or qualification.	7.1.1 Undertake more complex professional activities involving the selection and application, based on professional judgement of new and existing techniques and methodologies requiring the exercise of professional independence combine with competence derived from extensive experience and/or additional study. 7.1.2 Undertake supervisory responsibilities which may include on the job training, staff assessment and performance counselling in relation to staff in lowel level positions. 7.1.3 Operate and be accountable for the quality of output of a section or function within the school.

Level 2	Level3	Level4	Level5	Level 6	Level 7
2.2.1 Provide science program assistance where limited discretion and judgement are involved 2.2.2 Maintain science equipment, materials and specimens not requiring a depth of knowledge or technical skills. 2.2.3 Under the direct supervision of an academic staff member(s) prepare and maintain laboratory teaching areas including routine setting up and dismantling of items of equipment for use in experimental, observational and teaching activities 2.2.4 Assist in the demonstration of experiments and scientific equipment under the direct supervision of an academic staff member(s).	 3.2.1 Provide science program assistance where some discretion and judgement are involved. 3.2.2 Assist in the design/demonstration of experiments under supervision of an academic staff member(s) where some discretion and judgement are involved. 3.2.3 Under direction, prepare, maintain, organize, set-up and dismantle equipment and materials for routine experiments or student projects and dispose of waste materials. 3.2.4 Peer assistance and/or guidance may be provided for other assistants in a laboratory. 3.2.5 Under direction and within existing routines, methods and procedures, prepare, maintain and dispense stock solutions, simple chemical mixtures and compounds, cultures or similar materials. 	 4.2.1 Responsibility for and/or training of subordinate staff in limited areas may be required. 4.2.2 Design and demonstrate experiments, within a variety of routines and procedures, under the supervision of an academic staff members(s) where discretion and judgement are required. 4.2.3 Where there is complexity in the range and choice of action and judgement are required: prepare, maintain and dispense solutions, chemical mixtures, compounds and cultures; prepare, maintain, organise, setup and dismantle equipment and material for experiments. 	5.2.1 Apply a range of technical and other skills involving the self-directed application of knowledge gained through formal studies/qualifications applicable to this level or knowledge and experience that are determined by the employer as necessary to successfully carry out the duties of the position. This may include: designing laboratory experiments; and appropriate responsibility for the application of workplace health and safety requirements in the laboratory	 6.2.1 Apply a range of professional knowledge gained through successful completion of an appropriate undergraduate degree. 6.2.2 Operate (at a level consistent with the qualifications required) a laboratory. This may (or may not) include responsibility for the supervision, monitoring and training of staff in lower level positions. 6.2.3 Administer the allocation and monitoring of resources in the laboratory. 6.2.4 Support employees reporting to the position in policies to be followed, methods to be used and standards to be observed. 6.2.5 Provide professional advice to staff and students in the officer's area of expertise. 6.2.6 Formulate procedural policy and guidelines in the employee's area of responsibility; submit recommendations for decision and prepare supporting statements as necessary. 	7.2.1 Apply a range of professional knowledge gained through successful completion of an appropriate undergraduate degree and post graduate qualifications and/or other professional development and/or industry experience. 7.2.2 Responsibility for the operation of a laboratory which provides complex and varied services. This may (or may not) include responsibility for the supervision, monitoring and training of professional staff and staff in lower level positions. 7.2.3 Manage the allocation and monitoring of resources in the laboratory. 7.2.4 Responsibility for direction and support of employees reporting to the position in policies to be followed, methods to be use and standards to be observed. 7.2.5 Provide subject matter expertise and/or policy advice across a range of programs of activities undertaken by the organizational area. Formulate policies and provide specialist advice on policy development to senior management.

Administration Employees

Level 2	Level 3	Level4	Levels	Level6	Level7
 2.3.1 Use keyboard skills to produce a document from written text using a standard format. 2.3.2 Receive and deal with enquiries within well established routines,, including the provision of general information and assistance to the public, parents, students and other employees. 2.3.3 Perform a range of general clerical duties at a basic level, for example, filling, handling mail, maintaining records. 	3.3.1 Carry out a wide range of secretarial and clerical duties at an advanced level, including shorthand, typing, word processing and maintaining manual and computerized records. 3.3.2 Respond to enquiries from staff, students, parents and the general public and address issues in accordance with routines, methods and procedures. 3.3.3 Enter financial data into computer and prepare financial and management reports for review and authorization by senior management.	 4.3.1 Provide administrative support to senior management of a school where discretion and judgement are required, including: taking minutes; shorthand; organizing appointments and diaries; initiating and handling correspondence (which may include confidential correspondence); monitoring telephone calls; and establishing and/or maintaining working filing systems. 4.3.2 Within a variety of routines, methods and procedures apply inventory and purchasing control procedures, prepare monthly summaries of debtors and creditors ledger transactions and reconcile these. 4.3.3 Apply knowledge of advanced functions of computer software packages and to manage data i.e. modify fields of information, develop new databases or spreadsheet models; or graph previously prepared spreadsheets. 	 5.3.1 Provide executive support to senior management and associated committees concerning designated aspects of school management. 5.3.2 Direct and supervise the work of administrative/clerical and/or other staff. 5.3.3 Under broad guidance, supervise the operations of the school's office and other administrative activities, in the areas of enrolment, equipment and statistical staffing returns. 5.3.4 Under broad guidance, supervise the operations of the school's processes and activities in relation to overseas students. This may include: enrolment; family liaison; and placement. 	 6.3.1 Operate and be responsible for an autonomous section and all its operations. 6.3.2 Provide professional advice to staff and students in the officer's area of expertise. 6.3.3 Monitor and analyse regular management information, such as staffing and financial resource usage; ensure that associated information systems are maintained and that regular reports are provided to management. 	 7.3.1 Supervise staff including implementation and participation in induction, training, review, counselling and appraisal 7.3.2 Manage the work of administrative officers and other staff, assigning and outlining the work, advising on administrative problems, and revising work for accuracy and adequacy. 7.3.3 Identify policies and procedures requiring review or redevelopment, and define relevant issues.

Administration Employees cont

- 2.3.4 Operate within wellestablished routines, office equipment, such as, computer, photocopier, facsimile, binding machine, guillotine, laminator, franking machine, calculators, switchboard, etc.
- 2.3.5 Under the direct supervision of the principal or nominee, contact parents, students and/or others in relation to school attendance and related matters.
- 2.3.6 Under the direct supervision of the principal or nominee assist with the arrangement of group meetings, morning teas, meetings of parents and external parties, parent/teacher nights etc.
- 2.3.7 Carry out minor cash transactions including receipting, balancing and banking.

- 3.3.4 Prepare and process payroll transactions within routines, methods and procedures.
- 3.3.5 Within routines, methods and procedures: provide administrative support to senior management; arrange appointments and diaries; and prepare correspondence.
- 3.3.6 Within routines. methods and procedures. prepare and dispatch statements to debtors and payments to creditors, follow up on unpaid accounts; prepare bank reconciliations and reconcile accounts to balance: maintain wage and salary records.
- 3.3.7 Maintain petty cash float and expenses for accounting purposes

- 4.3.4 From verbal or rough handwritten instructions; answer non-standard executive correspondence, prepare papers, briefing notes, or other written material.
- 4.3.5 Utilizing a variety of routines, methods and procedures, calculate and maintain wage and salary records; perform routine classification determinations; and process resignations, retirements and redundancies in accordance with relevant award entitlements.
- 4.3.6 Within a variety of routines, methods and procedures provide significant assistance in the preparation of: financial information to trial balance; budgets; cash flow records; balance sheets; trading accounts; cash management analysis; FBT and entity disclosure requirements.
 NOTE an employee is not required to perform all duties listed to satisfy this skill descriptor.
- 4.3.7 Train staff classified at lower levels by means of personal instruction and demonstration.
- 4.3.8 Within a variety of routines, methods and procedures provide significant assistance in the enrolment, family liaison and placement of overseas students.

- 5.3.5 Prepare the accounts of the school to operating statement stage and assist in the formulation of period and year-end entries.
- 5.3.6 Provide advice requiring knowledge of policies and/ or the interpretation of rules or regulations within their area of operation. Assist in developing policy and procedures relating to their work area and identifying future trends.
- 5.3.7 Under broad guidance supervise the administration of specialized salary and payroll requirements, which may include: eligible termination payments, superannuation trust deed requirements, redundancy calculations or workers' compensation claims.
- 5.3.8 Prepare for senior management financial reports relating to the employee's area of responsibility.

- 6.3.4 Provide financial, policy, or planning advice which may include providing reports, statistical surveys and advice on regulations and procedures.
- 6.3.5 Monitor expenditure against a budget ata school level, draft financial forecasts / budgets at organizational level and / or prepare complex financial reports.
- 6.3.6 Administer programs with a range of tasks such as advice on financial implications, interpretation of information, assistance and advice concerning complex issues.
- 6.3.7 Prepare correspondence which is complex, original and which initiates or responds to new cases or situations.

- 7.3.4 Provide written reports to the school executive on complex matters, suggesting alternative courses of action and analysing the implications of each alternative.
- 7.3.5 Provide financial, policy and planning advice and investigate, interpret or evaluate information for the guidance of staff or clients.
- 7.3.6 Be substantively involved in the construction of annual and forward planning school budgets
- 7.3.7 Manage theoperations of a discrete organizational area, program or administrative function.

Administration Employees cont

- 2.3.8 Monitor and maintain stock levels of stationery/materials for office/department within established parameters including reordering.
- 2.3.9 Within well
 established routines,
 sort, prepare and
 record documents
 (e.g. invoices,
 cheques,
 correspondence) on
 a daily basis; file
 such documents in
 the appropriate
 system.
- 2.3.10 Within well
 established routines,
 receive and
 distribute incoming
 mail collect outgoing
 mail, maintain mail
 registers and records
 and collate and
 dispatch documents
 for bulk handling.
- 2.3.11 Perform, within wellestablished routines, tasks associated with the mass production of printed material including collating, stapling, binding, folding, cutting, etc.

- 3.3.8 Assist in the preparation of internal and external publications.
- 3.3.9 Assist in the enrolment function including handling initial enquiries and arranging interviews.
- 3.3.10 Under supervision, prepare Government and Statutory Authority returns for authorization by senior management.
- 3.3.11 Use software application packages for personal computers to create database file structures; and spreadsheets/work sheets.
- 3.3.12 Under direction and within routines, methods and procedures: draft agenda for meetings; assemble supporting documents for informal meetings; take and produce minutes.
- 3.3.13 Draft and type routine correspondencefrom brief oral or written instructions. Respond to requests for information including drafting routine correspondence in reply.

- 5.3.9 Apply a knowledge of relevant industrial awards and agreements and occupational health and safety requirements. Provide general advice to staff in these areas.
- 5.3.10Original writing of promotional and advertising material.
- 5.3.11Coordination of, and participation in, marketing activities.
- 5.3.12Design of promotional and marketing plans.
- 5.3.13Responsibility for liaison with media.

- 6.3.8 Formulate procedural policy and guidelines in the employee's area of responsibility; submit recommendations for decision and prepare supporting statements as necessary.
- 6.3.9 Direct and support employees reporting to the position in policies to be followed, methods to be used and standards to be observed.
- 6.3.10 Advise and assist in the preparation of the school budget.
- 6.3.11 Provide executive support to Principals and senior management.
- 6.3.12 Provide advice or make recommendations requiring detailed knowledge of policies, and/or the interpretation of rules or regulations within established guidelines, relating to a major function of the organizational work areas.
- 6.3.13 Supervise staff including participation in induction, training, review, counselling and appraisal and providing feedback on performance.

- 7.3.8 Provide subject matter expertise and/orpolicy advice across a range of programs or activities undertaken by the organizational area, formulate policies and provide specialist advice on policy formulation to senior management.
- 7.3.9 Undertake high level research, review or investigations including the preparation of reports and associated papers to provide advice to the school on the operational and/or future directions of the employee's section and to contribute to the development of that section in the educational context of the school.
- 7.3.10 Prepare papers, investigate and present information with recommendations for decision by senior officers.

- 3.3.14 Maintain established central filing / records systems in accordance with routines, methods and procedures. This would include: creating and indexing new files, retrieving records; distributing files within the school as requested, monitoring file locations and identifying and processing inactive and closed files.

 3.3.15 Maintain a store through
- 3.3.15 Maintain a store through such duties as participation in ordering and issue of expendable stores, recording of stock levels, maintaining records of equipment distribution, delivery dockets, invoices and payment vouchers and responsibility for keys.
- 3.3.16 Make and record appointments on behalf of another and, where necessary, resolve involved appointment scheduling problems.
- 3.3.17 Make travel and accommodation bookings in line with a given itinerary.
- 3.3.18 Within routines, methods and procedures carry out liaison between the school, the student and the student's family where some discretion and judgement are involved.

- 6.3.14 Develop systems and procedures for implementation in accordance with school policy.
- 6.3.15 Provide financial, policy and planning advice and investigate, interpret or evaluate information for the guidance of staff or clients.
- 6.3.16 Original writing of promotional and advertising material requiring significant discretion and judgement concerning content and design.
- 6.3.17 Management of, and participation in, marketing activities.
- 6.3.18 Design of promotional and marketing plans requiring initiative in the application of professional practices.
- 6.3.19 Responsibility for representing the school in the media.

Level 2	Level 3	Level 4	Level 5	Level 6	Level 7
 2.4.1 Process basic transactions such as issues and returns, produce overduelists, entry of orders in a computerised system, perform stock takes, entering of accession information into computer. 2.4.2 Operate and demonstrate the use of audio-visual equipment where there is limited complexity. 2.4.3 Maintain a booking system for equipment use and for the organisation of repairs and replacement of equipment. 2.4.4 Within well-established routines, methods and procedures, record audio/video programs and maintain a catalogue system of such recordings. 2.4.5 Perform a range of general duties at a basic level, for example, minor book repairs, photocopying and shelving. 2.4.6 Receive and deal with initial requests for information from library clients. 	3.4.1 Search and verify bibliographical data where some discretion and judgement are involved. 3.4.2 Copy catalogue books, magazines, journals and recorded material where some discretion and judgement are involved. 3.4.3 Maintain circulation systems where some discretion and judgement are involved. 3.4.4 Respond to enquiries from staff, students, parents and the general public and address issues in accordance with routines, methods and procedures. 3.4.5 Assist in the demonstration of complex audio visual or computer equipment under supervision of academic staff member(s) where some discretion and judgement are involved.	4.4.1 Responsibility for and/or training of subordinate staff in limitedareas may be required 4.4.2 Within a variety of routines and procedures and with a depth of knowledge in some areas: demonstrate to staff and students the use of complex audio visual or computer equipment; or monitor performance of and carry out repairs to specialised equipment.	5.4.1 Apply a range of technical and other skills involving the self-directed application of knowledge gained through formal studies/qualifications applicable to this level or knowledge and experience that are determined by the employer as necessary to successfully carry out the duties of the position. This may include: independent and original cataloguing and classification following precedents and standards; monitoring the performance of, and carrying out repairs to, specialised equipment; and developing the framework for and providing the instruction to students (within a structured learning environment) under the general supervision of an academic staff member(s).	6.4.1 Apply a range of professional knowledge gained through successful completion of an appropriate undergraduate degree. 6.4.2 Operate (at a level consistent with the qualifications required) a library/resource centre. This may (or may not) include responsibility for the supervision, monitoring and training of staff in lower level positions. 6.4.3 Administer the allocation and monitoring of resources in the library/resource centre. 6.4.4 Support employees reporting to the position in policies to be followed, methods to be used and standards to be observed. 6.4.5 Provide professional advice to staff and students in the officer's area of expertise. 6.4.6 Formulate procedural policy and guidelines in the employee's area of responsibility;	7.4.1 Apply a range of professional knowledge gained through successful completion of an appropriate undergraduate degree and post graduate qualifications and/or other professional development and/or industry experience. 7.4.2 Responsibility for the operation of a library/resource centre which provides complex and varied services. This may (or may not) include responsibility for the supervision, monitoring and training of professional staff and staff in lower level positions. 7.4.3 Manage the allocation and monitoring of resources in the library/resource centre. 7.4.4 Responsibility for direction and support of employees reporting to the position in policies to be followed, methods to be used and standards to be observed. 7.4.5 Provide subject matter expertise and/or policy advice across a range of programs or activities undertaken by the organizational area, formulate policies and provide specialist advice on

2.4.7 Under the direct supervision of an academic staff member(s), assist in the demonstration of routine library operations and procedures.		submit recommendations for decision and prepare supporting statements as necessary.	policy formulation to senior management.
2.4.8 Under the direct supervision of an academic staff member(s), assist in the supervision of students in the library.			
2.4.9 Under direct supervision of a higher level officer or members of the academic staff prepare and clear away materials for display/use in classrooms or libraries.)			

Computer/ICT Employees

Level 2	Level 3	Level 4	Level 5	Level6	Level ?
2.5.1 Use keyboard skills to produce a document from written text using a standard format 2.5.2 Operate within well established routines, office equipment, such as, computer, photocopier, facsimile, binding machine, guillotine, laminator, franking machine, calculators, switchboard, etc.	3.5.1 Use software application packages for personal computers to create database file structures; and spreadsheets/work sheets.	4.5.1 Within a variety of routines, methods and procedures, maintain the hardware and software components of a computer network and provide user support. 4.5.2 Responsibility for and/or training of subordinate staff in limited areas may be required.	5.5.1 Apply a range of technical and other skills involving the self-directed application of knowledge gained through formal studies/qualifications applicable to this level or knowledge and experience that are determined by the employer as necessary to successfully carry out the duties of the position. This may include: Assisting with systems analysis and design in relation to the development and maintenance of computer systems; and assisting with application programming (e.g. modification of package systems; and investigation of malfunctions in operational programs).	6.5.1 Operate and be responsible for the computing section of the school and all its operations. 6.5.2 Perform non-routine professional tasks governed by procedures or guidelines. When such constraints the employee is responsible for the independent performance of such functions. 6.5.3 Provide financial, policy and planning advice and investigate, interpret or evaluate information for the guidance of staff or management in the computing area. 6.5.4 Be responsible for the development of software, hardware or applications systems based on the use of current computer techniques. 6.5.5 Be responsible for the development of computer systems, and recommend changes and improvements in systems where appropriate.	7.5.1 Operate and be responsible for the computing section which provides complex and varied services to the school community including being responsible for the supervision, monitoring and development of other staff reporting to the position. 7.5.2 Research and examine likely long-term requirements for computer systems, suggest alternative plans and strategies and report on their feasibility. 7.5.3 Consult with departmental computer users to understand and meet the needs of the department and resolve problems concerning systems. 7.5.4 Investigate and design the implementation of computer systems to meet specific needs of work areas.

		6.5.6 Undertake maintenance programming tasks, including investigation and design requirements necessary to implement changes to existing systems. 6.5.7 Provide advice to the senior executive of the school on the operations/future directions of the section by utilising acquired knowledge and experience. 6.5.8 Carry out a range of tasks necessary to support and develop systems software or other support processes	7.5.5	of complex and varied tasks requiring the selection and application of new and existing techniques and methodologies necessary to support and develop systems software or other support processes.
--	--	--	-------	--

SCHEDULE 4 - CONDITIONS SPECIFIC TO SERVICES STAFF

Subject Matter			Clause No.		
PART 1 - APPLICATION					
	Title and Coverage				
PART 2 - EMPLOYER AND EMPLOYEES' DUTIES, EMPLOYMENT RELATIONSHIP AND RELATED ARRANGEMENTS					
Part-tim Casual Term T	mployment Categories				
PART 3 - HOURS OF WORK, BREAKS, PENALTY RATES, WEEKEND WORK AND OVERTIME					
Hours of Work					
1.1 Title and Coverage					
	(a)	This Schedule will be called the Services Staff Schedule (Schedule 4). Schedule will provide the terms and condition of employment for employees who are not: teachers; school officers; boarding superviso childcare; or nurses.			
	(b)	The Classifications and wages for employees covered by this Schedule contained in Schedule 5.	e are		
	(c)	Where a particular provision appears in the body of the Agreement, the provision will override (to the extent of any inconsistency) a provision this Schedule which deals with the same matter.			
1.2	Definitions				
1.2.1	The "Act" means the Fair Work Act 2009.				
1.2.2	"Union" means the Independent Education Union of Australia- Queensland and Northern Territory Branch and/or any other organisation which is registered pursuant to the Fair Work (Registered Organisations) Act 2009 and which is eligible to cover an employee to which this Agreement applies.				

1.2.3 "Green keeping and grounds employees" include all employees involved in

- preparation and maintenance of playing greens, sports grounds and courts and school grounds.
- 1.2.4 "Building maintenance employees" means employees who are engaged in the repair and renovation of buildings and structures necessitating the use of building tradespersons', or labourers' skills and/or tools

PART 2 - EMPLOYER AND EMPLOYEES' DUTIES, EMPLOYMENT RELATIONSHIP AND RELATED ARRANGEMENTS

2.1 Employment Categories

2.1.1 Employees (other than casuals) covered by this Agreement shall be advised in writing of their employment category upon appointment. In the case of casual employees such notification need only be supplied at the initial engagement and when that employee's employment status changes.

Employment categories are:

- (a) full-time;
- (b) part-time (as prescribed in clause 2.2); or
- (c) casual (as prescribed in clause 2.3);
- (d) trainee (as prescribed by clause 2.4);
- (e) term time (as prescribed by clause 1.4 in Schedule 3).

2.2 Part-time Employment

- 2.2.1 A part time employee means an employee who is employed on a continuing contract of employment who works (subject to this clause) fewer hours than a full time employee.
- 2.2.2 A part time employee will be employed as follows:
 - the spread of ordinary working hours shall be the same as those prescribed for full- time employees;
 - (b) the number of ordinary working hours in any one week shall not be less than 12 and shall not exceed 32;
 - (c) the spread of ordinary working hours shall be the same as those prescribed for full- time employees;
 - the ordinary daily working hours shall be worked continuously, excluding meal breaks, and shall not be less than 4 hours or more than 10 hours per day;
 - (e) an hours worked outside, or in excess of the ordinary working hours will be paid for at the overtime rate;
 - (f) part time employees shall be paid an hourly rate equal to the appropriate weekly rate divided by 38;Part time employees shall be entitled to a

proportionate amount of annual leave, sick leave, long service leave, bereavement leave and all public holidays as prescribed for weekly employees;

(g) all other provisions of this Agreement relevant to full-time employees shall apply to part-time employees.

2.3 Casual Employment

- 2.3.1 A casual employee is an employee who engaged as such, who is employed by the hour and who works less than 38 ordinary hours per week.
- 2.3.2 A casual employee will be employed as follows:
 - (a) A casual employee will be paid 25% in addition to the applicable rates of wages prescribed for a similar employee;
 - (b) A casual employee will be paid for a minimum of two hours for each engagement.

2.4 Term Time Employment

Employees may be employed as term time employees in accordance with clause 1.4 in Schedule 3.

2.5 Anti-Discrimination

- 2.5.1 It is the intention of the parties to this Agreement to prevent and eliminate discrimination, as defined by the *Anti-Discrimination Act 1991* and the *Fair Work Act 2009*, which includes:
 - (a) discrimination on the basis of sex, marital status, family responsibilities, pregnancy, parental status, age, race, impairment, religion, political belief or activity, trade union activity, lawful sexual activity and association with, or relation to, a person identified on the basis of any of the above attributes;
 - (b) sexual harassment; and
 - (c) racial and religious vilification.
- 2.5.2 Accordingly, in fulfilling their obligations under the grievance and dispute settling procedure in clause 3.1, the parties to this Agreement must take reasonable steps to ensure that neither the Agreement provisions nor their operation are directly or indirectly discriminatory in their effects.
- 2.5.3 Under the *Anti-Discrimination Act 1991* it is unlawful to victimise an employee because the employee has made or may make or has been involved in a complaint of unlawful discrimination or harassment.
- 2.5.4 Nothing in clause 2.6 is to be taken to affect:
 - (a) any different treatment (or treatment having different outcomes) which is specifically exempted under the Anti-Discrimination Act 1991;
 - (b) an employee, employer or registered organisation pursuing matters of discrimination, including by application to the Human Rights and Equal Opportunity Commission/Anti- Discrimination Commission Queensland.

PART 3 - HOURS OF WORK, BREAKS, PENALTY RATES, WEEKEND WORK AND OVERTIME

3.1 Hours of Work

3.1.1 Ordinary Hours of Work

- (a) Subject to clause 3.1.3, the ordinary hours of work for all employees shall be an average of 38 per week, to be worked on one of the following bases:
 - (i) 38 hours within a work cycle not exceeding 7 consecutive days; or
 - (ii) 76 hours within a work cycle not exceeding 14 consecutive days; or
 - (iii) 114 hours within a work cycle not exceeding 21 consecutive days; or
 - (iv) 152 hours within a work cycle not exceeding 28 consecutive days.
- (b) The ordinary hours of work will be worked on not more than 5 consecutive days in a week. The ordinary hours of work will be worked continuously, except for meal breaks.
- (c) The ordinary working hours will be worked on the following days and between the following times:
 - (i) Building and building maintenance employees, Retail employees: Monday to Friday inclusive, between 6.00 a.m. and 6.00 p.m.
 - (ii) Green keeping and Grounds employees: Monday to Sunday inclusive, between 6.00 a.m. and 6.00 p.m.
 - (iii) Catering and domestic employees, Employees driving motor vehicles and involved in associated duties: Monday to Sunday inclusive
 - (iv) Cleaners, caretakers, security employees: Monday to Sunday inclusive between 6.00 a.m. and 6.00 p.m.
- (d) Work done outside of the ordinary hours prescribed in this clause (clause 4.1.1) will be paid at overtime rates.
- (e) All employees will work their ordinary working hours in accordance with a roster. That roster will not be changed except by seven (7) days' notice. The roster shall be displayed in a conspicuous place or places accessible to the employees concerned.
- (f) The maximum number of ordinary hours of work shall not exceed 10 hours any day:

3.1.2 Working of 38 hour week

- (a) The 38 hour week shall be worked in one of the following ways, most suitable to the particular employer, after consultation with, and giving reasonable consideration to the wishes of the employees concerned:-
 - (i) by employees working less than 8 ordinary hours each day; or
 - (ii) by employees working less than 8 ordinary hours on one or more days

- each work cycle; or
- (iii) by fixing one or more work days on which all employees will be off during a particular work cycle; or
- (iv) by rostering employees of on various days of the week during a particular work cycle, so that each employee has one work day off during that cycle.
- (b) Subject to clause 3.1.2(a), employees may agree that the ordinary hours of work are to exceed 8 on any day, thus enabling more than one work day to be taken off during a particular cycle.
- (c) Notwithstanding any other provision clause 3.1.2, where the arrangement of ordinary hours of work provides for a rostered day off, the employer and the majority of employees concerned, may agree to accrue up to a maximum of 5 rostered days off. Where such an agreement has been reached, the accrued rostered days off shall be taken within 12 calendar months of the date of which the first rostered day off was accrued. Consent to accrue rostered days off shall not be unreasonably withheld by either party.
- (d) When the ordinary working cycle provides for a rostered day off, the rostered day off shall not fall on a public holiday, but shall be on the ordinary working day immediately before or immediately after the public holiday, or deferred in accordance with clause 3.1.2(c).
- (e) Different methods of working of the 38 hour week may apply to individual employees, groups or sections of employees in the organisation concerned.
- (f) Ordinary hours for all employees shall be paid on the basis of not more than 38 per week, on an averaged basis according to the work cycle, notwithstanding that in excess of 38 ordinary hours may be worked to maximise leisure time off in accordance with this clause (clause 3.1.2).

3.1.3 Procedures for Enterprise Level Discussions

- (a) The employer and all employees concerned in each establishment shall consult over the most appropriate means of working a 38 hour week.
- (b) The objective of such consultation shall be to reach agreement on the method of working the 38 hour week in accordance with clause 3.1.2.
- (c) The outcome of such consultation shall be recorded in writing.
- (d) In cases where agreement cannot be reached as a result of consultation between the parties, either party may request the assistance or advice of their relevant employee or employer organisation.
- (e) Notwithstanding the consultative procedures outlined above, and notwithstanding any lack of agreement by employees, the employer shall have the right to make the final determination as to the method by which the 38 hour week is implemented or worked from time to time.
- (f) Upon giving 7 days' notice or such shorter period as may be mutually agreed upon, the method of working the 38 hour week may be altered, from time to time, following negotiations between the employer and employees concerned, utilising the provisions of clause 3.1.

3.2 Meal Breaks

- 3.2.1 Employees are entitled to a meal break of between thirty minutes and one hour when they work for five or more hours on any one day. The meal break will be taken between the 4th and 6th hours from the commencement of work.
- 3.2.2 All work done during the recognised meal break shall be paid for at the rate of double time, such payment to continue until a meal break has commenced. Such meal break shall be of the duration prescribed for the ordinary hours of work by clause 3.2.1.
- 3.2.3 Hot water shall be provided by the employer for meal breaks and rest pauses.
- 3.2.4 Employees who are required to continue working for more than one and a-half hours beyond their ordinary finishing time shall be entitled to take a 30 minute paid meal break and shall be provided with an adequate meal by the employer or paid an allowance of \$16.53 in lieu thereof. Where an employee has provided a meal because of receipt of notice to work overtime and such overtime is not worked that employee shall be paid \$16.53 for any meal so provided.

3.3 Rest Pauses

- 3.3.1 Full time employees will receive a rest pause of 10 minutes duration in the 1st and 2nd half of each day worked. The rest pauses may be combined into one 20 minute rest pause. Where rest pauses are combined the working day shall be divided into 3 approximately equal periods of work.
- 3.3.2 Employees other than full-time are entitled to a rest break of 10 minutes for each period of 3 hours worked (excluding meal breaks) with a maximum of 2 rest breaks per shift. Such rest pauses shall be taken at such times as will not interfere with continuity of work where continuity is necessary.
- 3.3.3 Rest pauses shall be taken in the employer's time.
- 3.3.4 Rest pauses shall be taken at times to suit the convenience of the employer so as not to interfere with the continuity of work where continuity is necessary.

3.4 Overtime

- 3.4.1 All time worked in excess of or outside of the ordinary hours of work will be deemed overtime. Such hours may be compensated using either this clause (clause 3.4 Overtime) or clause 3.5 (time off in lieu of overtime).
- 3.4.2 All overtime in any one day, except as otherwise provided, will be paid for at one and a half times the ordinary rate for the first 3 hours and double time thereafter. All overtime worked on a Sunday will be at the rate of double time.
- 3.4.3 An employee recalled to work overtime after leaving the employee's business premises (whether notified before or after leaving the premises) will be paid for a minimum of 3 hours' work at the appropriate rate.
- 3.4.4 All overtime worked on a Saturday or on a Sunday shall be subject to a minimum payment as for 2 hours worked for each occasion that an employee is required to attend for duty. This minimum payment shall not be applicable where overtime is worked continuously with ordinary working hours on a Saturday.
- 3.4.5 Where an employee is recalled from home to work overtime, the employee shall

be paid for the time so worked at the rate of double time, with a minimum payment as for 3 hours' work in respect of each such recall.

3.4.6 When calculating overtime payments, any part of a-half of an hour that is worked on any one day shall be paid for as a full half of an hour. Where an employee has worked so much overtime between the termination of the employee's ordinary work on one day, and the commencement of the employee's ordinary work on the next day that the employee has not had at least 10 consecutive hours off duty between those times the employee will be released after completion of such overtime until the employee has had ten consecutive hours off duty without loss of pay for ordinary working time occurring during such absence.

If on the instructions of the employer such an employee resumes or continues work without having had such 10 consecutive hours off duty, the employee will be paid double rates until the employee is released from duty for a period often consecutive hours off duty without loss of pay for ordinary working time occurring during such absence.

3.5 Time off in lieu of Overtime

- 3.5.1 An employee and an employer may agree that the employee will receive paid time off in lieu of overtime. Any such agreement must be recorded in writing.
- 3.5.2 Time off in lieu of overtime shall be at the equivalent of the number of hours of ordinary pay that the employee would have received for such overtime.
- 3.5.3 Where there is written agreement between an Industrial Organisation and the employer time off in lieu of overtime may be accrued for a period longer that 12 months and/or a quantum in excess of 38 hours may be accrued.
- 3.5.4 Any accrued time off in lieu that is outstanding after 12 months (in the absence of written agreement between the Industrial Organisation and the employer) or at the time of termination of employment, for any reason, by either party, shall be paid out at the employee's ordinary time rate of pay.

SCHEDULE 5 - CONDITIONS SPECIFIC TO TAFE TRAINERS

5.1. Classifications - TAFE Trainer

- 5.1.1 On appointment a TAFE Tutor / Trainer will be placed on a salary level commensurate with the minimum salary for their qualifications and experience.
- 5.1.2 TAFE Tutors who have not taught before will commence at Foundation Educator and progress to a maximum of Senior Tutor 2 of the TAFE Trainers' classification structure specified in Schedule One.
- 5.1.3 TAFE Trainers who have not taught before will commence at Trainer 1 and progress to a maximum of Senior Trainer 3 of the TAFE Tutor / Trainers' classification structure specified in Schedule One.
- 5.1.4 Subject to the continuing satisfactory conduct, diligence and performance of a tutor / trainer and the acquisition and utilisation of skills and knowledge through experience, progression from one salary level to the next will occur on the completion of a year of full-time experience or equivalent part-time experience.

5.2 Hours of Work

- 5.2.1. The ordinary hours of work for a full-time TAFE Tutor / Trainer are 38 hours per week.
- 5.2.2 Such ordinary hours of work shall be worked continuously (except for meal breaks) between 8:00am and 4:00pm on Mondays to Fridays inclusive, during school terms only.
- 5.2.3 The normal starting and finishing times with the span of ordinary hours shall be established at the point of engagement and may only be varied by the employer with two (2) weeks' notice, or shorter period by mutual agreement.

5.3 Rest Pauses

- 5.3.1 Full-time employees shall receive a paid rest pause of 10 minutes' duration in the first half and the second half of each day worked.
- 5.3.2 Part time and casual Employees who work between four and six consecutive ordinary hours are entitled to a paid rest pause of 10 minutes' duration. Employees who work more than six consecutive ordinary hours (excluding the unpaid meal break) on any one day are entitled to a 10-minute rest pause in both the first half and the second half of the period worked.
- 5.3.3 Such rest pauses shall be taken at such times as will not interfere with continuity of work where continuity is necessary.
- 5.3.4 Notwithstanding the foregoing, where the employer and the employees agree, rest pauses may be combined for an employee who works more than six consecutive hours.

5.4 Meal Break

5.4.1 An employee is entitled to an unpaid meal break of not less than half an hour and not more than one hour per working day if they work more than five hours.

5.5 Annual leave

- 5.5.1 TAFE Tutors / Trainers are entitled to annual leave in accordance with Part 2-2 of the Fair Work Act 2009.
- 5.5.2 TAFE Tutors / Trainers must take annual leave during non-term weeks and, unless agreed otherwise between the Employer and an Employee, annual leave must be taken in the four-week period immediately following the final term week of the school year.
- 5.5.3 An employee who has served throughout the school year is entitled to a leave loading of 17.5% on four weeks' annual leave. The loading will normally be paid:
 - at the time that the employee is paid annual leave or pro rata annual leave;
 or
 - on the termination of employment by either party.
- 5.5.4 Leave loading is to be calculated using the following formula:

[Weekly salary x 4 x 17.5%] x term weeks worked by the employee in that school year

Total term weeks in that school year

For example, in the case of an employee with a weekly salary of \$1000 on termination of employment (or at the end of the final term week in the school year) who was employed at the school for 20 of the 38 term weeks in that school year, the calculation will be as follows:

 $1000 \times 4 \times 17.5\% = 700$

 $$700 \times 20/38 = 368.42

SCHEDULE 6 - CONDITIONS SOCIAL AND COMMUNITY WORKERS

6.1 Classifications – Social and Community Workers

- 6.1.1. On appointment a Social or Community Worker will be placed on a salary level commensurate with the minimum salary for their qualifications and experience.
- 6.1.2. Social and Community Workers without relevant work experience will commence at Level 3 step 1 and progress to a maximum of Level 7 Step 1 of the Social and Community Workers classification structure specified in Schedule One.
- 6.1.3. Subject to the continuing satisfactory conduct, diligence, and performance of a Social and Community Worker and the acquisition and utilisation of skills and knowledge through experience, progression from one salary level to the next will occur on the completion of a year of full-time experience or equivalent part-time experience.

6.2. Hours of Work

- 6.2.1. The ordinary hours of work for a full-time Social and Community Worker are 38 hours per week.
- 6.2.2. Such ordinary hours of work shall be worked continuously (except for meal breaks) between 8:00am and 4:00pm on Mondays to Fridays inclusive, during school terms only.
- 6.2.3. The normal starting and finishing times with the span of ordinary hours shall be established at the point of engagement and may only be varied by the employer with two (2) weeks' notice, or shorter period by mutual agreement.

6.3. Rest Pauses

- 6.3.1. Full-time employees shall receive a paid rest pause of 10 minutes' duration in the first half and the second half of each day worked.
- 6.3.2. Employees other than full-time are entitled to a rest break of 10 minutes for each period of 3 hours worked (excluding meal breaks) with a maximum of 2 rest breaks per shift. . Such rest pauses shall be taken at such times as will not interfere with continuity of work where continuity is necessary.
- 6.3.3 Such rest pauses shall be taken at such times as will not interfere with continuity of work where continuity is necessary.
- 6.3.4. Notwithstanding the foregoing, where the employer and the employees agree, rest pauses may be combined for an employee who works more than six consecutive hours.

6.4 Meal Break

6.4.1 An employee is entitled to an unpaid meal break of not less than half an hour and not more than one hour per working day if they work more than five hours.

6.5 Annual leave

- 6.5.1 Social and Community Workers are entitled to annual leave in accordance with Part 2-2 of the Fair Work Act 2009.
- 6.5.2 Staff must take annual leave during non-term weeks and, unless agreed otherwise between the Employer and an Employee, annual leave must be taken in the four-week period immediately following the final term week of the school year.

IN THE FAIR WORK COMMISSION

Fair Work Act 2009 (Cth) ("FW Act")

Matter number:

AG2021/8595

Employer:

Parklands Christian College Ltd
T/A Parklands Christian College
(Employer)

Application:

Section 185 – Application for approval of a single enterprise agreement, namely the

Parklands Christian College Collective Enterprise Agreement 2021

(Agreement)

Authorised representative:

Gary Cully Principal

Undertaking-Section 190

For and on behalf of the Employer I, Gary Cully:

- 1. declare that I have:
 - authority to give this undertaking on behalf of the Employer,
 - b. sought the views of all bargaining representatives for this undertaking pursuant to s 190(4) of the FW Act,
- 2. understand that each undertaking is to be taken to be a term of the Agreement.
- 3. give the following undertaking/s with respect to the Agreement:
 - a. Shift work: School Officers, Services Staff and Social and Community workers:

The employer does not and will not employ shift workers in the above categories.

Services staff and school officers – part time overtime:
 Notwithstanding the provisions of the Agreement, the employer will pay overtime after eight (8) hours to these categories of employees. Further, Clause 21.3 Reasonable additional hours – part-time employees of the

Educational Services (Schools) General Staff Award 2020 will be deemed to form part of this Agreement.

c. Social and Community workers – part time and casual employees: Where the Agreement is silent on the terms and conditions for these categories of employees, the relevant provisions of the Educational Services (Schools) General Staff Award 2020 will apply to them.

d. Services Staff - TOIL:

The employer will pay out any untaken TOIL after six (6) months at the overtime rates and will pay out any untaken TOIL on termination at the overtime rates, as per Clause 21.4 of the Educational Services (Schools) General Staff Award 2020.

e. Public Holidays:

The employer will pay an employee, required to work on a public holiday, at the rate of 250% of the minimum hourly rate for ordinary hours worked.

f. Allowances:

Where the Educational Services (Schools) General Staff Award 2020 provides for an allowance at Clause 19, such allowances are deemed to be a part of this Agreement.

g. Higher Duties:

Notwithstanding of the Agreement, Clause 17.5 (Higher Duties) of the Educational Services (Schools) General Staff Award 2020 will be deemed to form part of this Agreement.

h. Part-time and casual (Post-Secondary Education):

Notwithstanding the provisions of the Agreement, TAFE Tutor/Trainers employed on a part-time or casual basis will receive the terms and conditions provided by the Educational Services (Post-Secondary Education) Award 2020 where the Agreement is silent, and such terms and conditions will be deemed to form part of this Agreement.

Higher Duties (Post-Secondary Education):

Notwithstanding the provisions of the Agreement the content of Clause 15.4 (Higher Duties) of the Educational Services (Post-Secondary Education) Award 2020 will be deemed to form part of this Agreement.

Date signed:	7 th December 2021	
For and on behalf of the Employer by:	Gary Cully	
[In accordance with s.190(5) of the FW Act]		
Signature:	51-	
Witness name:	BY JEFREY GRAY	
Witness signature:	OB Gray	